Contents lists available at ScienceDirect



Nuclear Inst. and Methods in Physics Research, A

journal homepage: www.elsevier.com/locate/nima



# Full Length Article

# Towards a new generation of solid total-energy detectors for neutron-capture time-of-flight experiments with intense neutron beams

J. Balibrea-Correa<sup>1</sup>, V. Babiano-Suarez<sup>50,3</sup>, J. Lerendegui-Marco<sup>1</sup>, C. Domingo-Pardo<sup>1</sup>, I. Ladarescu<sup>1</sup>, A. Tarifeño-Saldivia<sup>1</sup>, G. de la Fuente-Rosales<sup>1</sup>, B. Gameiro<sup>1</sup>, N. Zaitseva<sup>49</sup>, V. Alcayne<sup>2</sup>, D. Cano-Ott<sup>2</sup>, E. González-Romero<sup>2</sup>, T. Martínez<sup>2</sup>, E. Mendoza<sup>2</sup>, A. Pérez de Rada<sup>2</sup>, J. Plaza del Olmo<sup>2</sup>, A. Sánchez-Caballero<sup>2</sup>, A. Casanovas<sup>3</sup>, F. Calviño<sup>3</sup>, S. Valenta<sup>4</sup>, O. Aberle<sup>5</sup>, S. Altieri<sup>6,7</sup>, S. Amaducci<sup>8</sup>, J. Andrzejewski<sup>9</sup>, M. Bacak<sup>5</sup>, C. Beltrami<sup>6</sup>, S. Bennett<sup>10</sup>, A.P. Bernardes<sup>5</sup>, E. Berthoumieux<sup>11</sup>, R. Beyer<sup>12</sup>, M. Boromiza<sup>13</sup>, D. Bosnar<sup>14</sup>, M. Caamaño<sup>15</sup>, M. Calviani<sup>5</sup>, D.M. Castelluccio<sup>16,17</sup>, F. Cerutti<sup>5</sup>, G. Cescutti<sup>18,19</sup>, S. Chasapoglou<sup>20</sup>, E. Chiaveri<sup>5,10</sup>, P. Colombetti<sup>21,22</sup>, N. Colonna<sup>23</sup>, P. Console Camprini<sup>17,16</sup>, G. Cortés<sup>3</sup>, M.A. Cortés-Giraldo<sup>24</sup>, L. Cosentino<sup>8</sup>, S. Cristallo<sup>25,26</sup>, S. Dellmann<sup>27</sup>, M. Di Castro<sup>5</sup>, S. Di Maria<sup>28</sup>, M. Diakaki<sup>20</sup>, M. Dietz<sup>29</sup>, R. Dressler<sup>30</sup>, E. Dupont<sup>11</sup>, I. Durán<sup>15</sup>, Z. Eleme<sup>31</sup>, S. Fargier<sup>5</sup>, B. Fernández<sup>24</sup>, B. Fernández-Domínguez<sup>15</sup>, P. Finocchiaro<sup>8</sup>, S. Fiore<sup>16,32</sup>, V. Furman<sup>33</sup>, F. García-Infantes<sup>34,5</sup>, A. Gawlik-Ramikega<sup>9</sup>, G. Gervino<sup>21,22</sup>, S. Gilardoni<sup>5</sup>, C. Guerrero<sup>24</sup>, F. Gunsing<sup>11</sup>, C. Gustavino<sup>32</sup>, J. Heyse<sup>35</sup>, W. Hillman<sup>10</sup>, D.G. Jenkins<sup>36</sup>, E. Jericha<sup>37</sup>, A. Junghans<sup>12</sup>, Y. Kadi<sup>5</sup>, K. Kaperoni<sup>20</sup>, G. Kaur<sup>11</sup>, A. Kimura<sup>38</sup>, I. Knapová<sup>4</sup>, M. Kokkoris<sup>20</sup>, Y. Kopatch<sup>33</sup>, M. Krtička<sup>4</sup>, N. Kyritsis<sup>20</sup>, C. Lederer-Woods<sup>39</sup>, G. Lerner<sup>5</sup>, A. Manna<sup>17,40</sup>, A. Masi<sup>5</sup>, C. Massimi<sup>17,40</sup>, P. Mastinu<sup>41</sup>, M. Mastromarco<sup>23,42</sup>, E.A. Maugeri<sup>30</sup>, A. Mazzone<sup>23,43</sup>, A. Mengoni<sup>16,17</sup>, V. Michalopoulou<sup>20</sup>, P.M. Milazzo<sup>18</sup>, R. Mucciola<sup>25,44</sup>, F. Murtas<sup>45</sup>, E. Musacchio-Gonzalez<sup>41</sup>, A. Musumarra<sup>46,47</sup>, A. Negret<sup>13</sup>, P. Pérez-Maroto<sup>24</sup>, N. Patronis<sup>31,5</sup>, J.A. Pavón-Rodríguez<sup>24,5</sup>, M.G. Pellegriti<sup>46</sup>, J. Perkowski<sup>9</sup>, C. Petrone<sup>13</sup>, E. Pirovano<sup>29</sup>, S. Pomp<sup>48</sup>, I. Porras<sup>34</sup>, J. Praena<sup>34</sup>, J.M. Quesada<sup>24</sup>, R. Reifarth<sup>27</sup>, D. Rochman<sup>30</sup>, Y. Romanets<sup>28</sup>, C. Rubbia<sup>5</sup>, M. Sabaté-Gilarte<sup>5</sup>, P. Schillebeeckx<sup>35</sup>, D. Schumann<sup>30</sup>, A. Sekhar<sup>10</sup>, A.G. Smith<sup>10</sup>, N.V. Sosnin<sup>39</sup>, M.E. Stamati<sup>31,5</sup>, A. Sturniolo<sup>21</sup>, G. Tagliente<sup>23</sup>, D. Tarrío<sup>48</sup>, P. Torres-Sánchez<sup>34</sup>, E. Vagena<sup>31</sup>, V. Variale<sup>23</sup>, P. Vaz<sup>28</sup>, G. Vecchio<sup>8</sup>, D. Vescovi<sup>27</sup>, V. Vlachoudis<sup>5</sup>, R. Vlastou<sup>20</sup>, A. Wallner<sup>12</sup>, P.J. Woods<sup>39</sup>, T. Wright<sup>10</sup>, R. Zarrella<sup>17,40</sup>, P. Žugec<sup>14</sup>

<sup>1</sup> Instituto de Física Corpuscular, CSIC - Universidad de Valencia, Spain

- <sup>2</sup> Centro de Investigaciones Energéticas Medioambientales y Tecnológicas (CIEMAT), Spain
- <sup>3</sup> Universitat Politècnica de Catalunya, Spain
- <sup>4</sup> Charles University, Prague, Czech Republic
- <sup>5</sup> European Organization for Nuclear Research (CERN), Switzerland
- <sup>6</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Pavia, Italy
- <sup>7</sup> Department of Physics, University of Pavia, Italy
  <sup>8</sup> INFN Laboratori Nazionali del Sud, Catania, Italy
- <sup>9</sup> University of Lodz, Poland
- <sup>10</sup> University of Manchester, United Kingdom
- <sup>11</sup> CEA Irfu, Université Paris-Saclay, F-91191 Gif-sur-Yvette, France

 $^{12} {\it Helmholtz-Zentrum \ Dresden-Rossendorf, \ Germany}$ 

<sup>13</sup> Horia Hulubei National Institute of Physics and Nuclear Engineering, Romania

\* Corresponding author. E-mail address: javier.balibrea@ific.uv.es (J. Balibrea-Correa).

https://doi.org/10.1016/j.nima.2024.170110

Received 11 October 2024; Received in revised form 14 November 2024; Accepted 21 November 2024 Available online 30 November 2024 0168-9002/© 2024 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/bync-nd/4.0/).

- <sup>14</sup> Department of Physics, Faculty of Science, University of Zagreb, Zagreb, Croatia
- <sup>15</sup> University of Santiago de Compostela, Spain
- <sup>16</sup> Agenzia nazionale per le nuove tecnologie (ENEA), Italy
- <sup>17</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Bologna, Italy
- <sup>18</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Trieste, Italy
- <sup>19</sup> Department of Physics, University of Trieste, Italy
- <sup>20</sup> National Technical University of Athens, Greece
- <sup>21</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Torino, Italy
- 22 Department of Physics, University of Torino, Italy
- <sup>23</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Bari, Italy
- <sup>24</sup> Universidad de Sevilla, Spain
- <sup>25</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Perugia, Italy
- <sup>26</sup> Istituto Nazionale di Astrofisica Osservatorio Astronomico di Teramo, Italy
- <sup>27</sup> Goethe University Frankfurt, Germany
- 28 Instituto Superior Técnico, Lisbon, Portugal
- <sup>29</sup> Physikalisch-Technische Bundesanstalt (PTB), Bundesallee 100, 38116 Braunschweig, Germany
- <sup>30</sup> Paul Scherrer Institut (PSI), Villigen, Switzerland
- <sup>31</sup> University of Ioannina, Greece
- 32 Istituto Nazionale di Fisica Nucleare, Sezione di Roma1, Roma, Italy
- 33 Affiliated with an institute covered by a cooperation agreement with CERN <sup>34</sup> University of Granada, Spain
- <sup>35</sup> European Commission, Joint Research Centre (JRC), Geel, Belgium
- <sup>36</sup> University of York, United Kingdom
- 37 TU Wien, Atominstitut, Stadionallee 2, 1020 Wien, Austria
- 38 Japan Atomic Energy Agency (JAEA), Tokai-Mura, Japan
- <sup>39</sup> School of Physics and Astronomy, University of Edinburgh, United Kingdom
- 40 Dipartimento di Fisica e Astronomia, Università di Bologna, Italy
- <sup>41</sup> INFN Laboratori Nazionali di Legnaro, Italy
- 42 Dipartimento Interateneo di Fisica, Università degli Studi di Bari, Italy
- <sup>43</sup> Consiglio Nazionale delle Ricerche, Bari, Italy
- 44 Dipartimento di Fisica e Geologia, Università di Perugia, Italy
- <sup>45</sup> INFN Laboratori Nazionali di Frascati, Italy
- <sup>46</sup> Istituto Nazionale di Fisica Nucleare, Sezione di Catania, Italy
- <sup>47</sup> Department of Physics and Astronomy, University of Catania, Italy
- 48 Department of Physics and Astronomy, Uppsala University, Box 516, 75120 Uppsala, Sweden
- 49 Lawrence Livermore National Laboratory, 7000 East Avenue, Livermore, CA 94551, USA
- <sup>50</sup> Universidad de Valencia, Spain

# ARTICLE INFO

Keywords: Neutron capture γ-ray detectors High sensitivity Hybrid setup Stilbene-d12

# ABSTRACT

Challenging neutron-capture cross-section measurements of small cross sections and samples with a very limited number of atoms require high-flux time-of-flight facilities. In turn, such facilities need innovative detection setups that are fast, have low sensitivity to neutrons, can quickly recover from the so-called  $\gamma$ flash, and offer the highest possible detection sensitivity. In this paper, we present several steps towards such advanced systems. Specifically, we describe the performance of a high-sensitivity experimental setup at CERN n\_TOF EAR2. It consists of nine sTED detector modules in a compact cylindrical configuration, two conventional used large-volume  $C_6D_6$  detectors, and one LaCl<sub>3</sub>(Ce) detector. The performance of these detection systems is compared using  ${}^{93}Nb(n, \gamma)$  data. We also developed a detailed GEANT4 Monte Carlo model of the experimental EAR2 setup, which allows for a better understanding of the detector features, including their efficiency determination. This Monte Carlo model has been used for further optimization, thus leading to a new conceptual design of a  $\gamma$  detector array, STAR, based on a deuterated-stilbene crystal array. Finally, the suitability of deuterated-stilbene crystals for the future STAR array is investigated experimentally utilizing a small stilbene-d12 prototype. The results suggest a similar or superior performance of STAR with respect to other setups based on liquid-scintillators, and allow for additional features such as neutron-gamma discrimination and a higher level of customization capability.

#### 1. Introduction

Historically, innovation on detection techniques and advances in high-quality pulsed neutron beams have led to fascinating discoveries in stellar nucleosynthesis and to subsequent refinements of theoretical models of stellar structure and galactic-chemical evolution [1-8]. For  $(n, \gamma)$  cross section measurements in the energy range relevant for astrophysics scintillation detectors based on deuterated benzene (C<sub>6</sub>D<sub>6</sub>) are being extensively used at the most active pulsed time-offlight facilities in the world, like CERN n\_TOF (Switzerland) [9], JRC (Belgium) [10] and Back-n (China) [11]. To a large extent, the success of these liquid scintillators is based on the possibility of applying the pulse-height weighting technique (PHWT) to convert them into

total-energy detectors (TED) in combination with their low neutron sensitivity [12-16]. The PHWT technique enables systematic uncertainties of only a few percent in the cross section determination, compatible with the accuracy required by stellar models [6]. In practice, the technique can be applied to any kind of  $\gamma$ -ray detector, provided that the detection probability remains low enough to avoid registering twoor more gamma-quanta from the same neutron capture cascade. PHWT allows for large flexibility in the design of the detection apparatus and allows further system optimization for specific experimental conditions. In the past, these optimizations included the replacement of the first generation of TED detectors, C<sub>6</sub>F<sub>6</sub> [17], by C<sub>6</sub>D<sub>6</sub> as well as original housing and structural materials with carbon fibers to reduce the neutron sensitivity of the setup [18,19]. Still, until recently most neutron-capture experiments have utilized liquid scintillators of relatively large volume (≈500 mL up to 1 L). However, large detection volumes in close geometry are not suitable for high-flux neutron beams,

such as the one at n\_TOF EAR2 [20]. In particular, the detectors suffer from significant pile-up and dead-time effects due to a very high count rate that is difficult to manage [21]. These problems then require larger distance to the sample, severely limiting the signal-to-background ratio, and thus affecting the detection sensitivity. In order to deal with such limitations, the segmented total-energy detector sTED [22,23] was designed for coping with the extreme conditions, reducing the active volume of individual detector cells in exchange of dealing with the dead-time corrections [21].

The reduction in active volume of the  $\gamma$ -ray detectors has led the n\_TOF collaboration to devote significant efforts in recent years to developing new experimental setups aimed at increasing the sensitivity of the experiments while minimizing systematic errors that could affect the measurements. It is worth mentioning that in the framework of the n TOF collaboration, other research lines try to cope with other scenarios where high count rate is not an issue, demonstrating the possibility of combining the TED principle with  $\gamma$ -ray imaging techniques for background suppression [24,25]. Additional studies have also shown the possibility of using the PHWT with high-efficiency setups, which aim to reduce statistical uncertainties and lower angular distribution effects [26]. In this work, we report on the features of a high-sensitivity, state-of-the-art detection system for  $(n, \gamma)$  crosssection measurements at n\_TOF EAR2. The system consists of nine sTED detectors in a close cylindrical geometric configuration. In addition, two C<sub>6</sub>D<sub>6</sub> detectors and one LaCl<sub>3</sub>(Ce) detector are positioned at 135° and at greater distances from the sample. In practice, C<sub>6</sub>D<sub>6</sub> presents significant disadvantages: it is highly toxic, flammable, and carcinogenic, thus requiring careful handling and costly maintenance. Therefore its replacement with a new material that would maintain, or even improve the overall detector performance appears to be necessary. One of the most promising options is the use of high-purity deuterated stilbene crystals [27,28], which, when combined with the high-sensitivity detector arrangement, could serve as a replacement for C<sub>6</sub>D<sub>6</sub> detectors in the near future, as will be discussed in the second part of the manuscript.

The present article is structured as follows: Section 2 introduces briefly the concept of signal-to-background ratio and highlights the importance of this ratio for a  $(n, \gamma)$  experiment and nuclear data evaluations. Section 3 presents the 2022 state-of-the-art experimental setup for  $(n, \gamma)$  cross section measurements at CERN n\_TOF EAR2, specially designed to improve the sensitivity with respect to previous setups. Section 4 introduces a Monte Carlo (MC) model of the presented setup based on GEANT4 [29] that allows one to quantify the efficiency of the detectors and serves as a starting point for tests of additional changes in the detection setup. Section 5 envisions a new generation of stilbene-d12 based total-energy detectors for future experiments at n\_TOF EAR2 and Section 6 describes initial characterization for a small stilbene-d12 prototype. Finally, Section 7 presents the main conclusions and outlook of this work.

# 2. The importance of signal-to-background ratio for an experiment

In practical terms, the quality of a Time-of-Flight (ToF) based neutron capture cross-section experiment strongly depends on several ingredients, such as high-quality sample preparation [30], beam-time restrictions, and strict control of systematic uncertainties associated with background determination and normalization [31–33]. However, as with any counting experiment the detection process is governed by statistics [34], and the actual number of detected events is thus influenced by statistical fluctuations. In addition, physical events are often detected together with other non-negligible background sources. One critical aspect in the design of the experiment thus involves optimizing the signal-to-background ratio (s/b) of the experimental setup. In practical applications, this ratio is defined as the number of registered counts during the measurement ( $n_i$ ) divided by the number of background counts registered or estimated in the same region  $(n_b),$  normalized by the number of neutron pulses devoted to each configuration,  $N^\prime$  and N, respectively.

$$s/b = \frac{n_i}{n_b(N'/N)} \equiv \frac{n_i}{n_b\alpha} \tag{1}$$

In the case that the individual pulses would have different intensity, the normalization (N' and N) can be extended to the sum of the neutrons devoted for each configuration. Depending on the s/b value, a larger or smaller number of neutron pulses is required to observe a significant result. The s/b is closely related to the precision determining a neutron capture cross section and the attainable accuracy of the resonance parameters. In practice, for a counting experiment – where both  $n_i$  and  $n_b$  come from the Poisson distribution – the statistical significance for an existence of a signal above the background can be described [35,36] by standard statistical *p*-value that is given by

$$p\text{-value} \equiv \int_{\lambda}^{\infty} N(x) dx \tag{2}$$

where N(x) is the normal distribution with zero mean and unit variance and  $\lambda$ 

$$\lambda = \sqrt{2} \left[ n_i \log \left( \frac{\alpha + 1}{\alpha} \frac{n_i}{n_i + n_b} \right) + n_b \log \left( (\alpha + 1) \frac{n_b}{n_i + n_b} \right) \right]^{1/2}.$$
 (3)

Given a significance level, the *p*-value can be use to assess whether the observed data correspond to a background fluctuation or a nonzero cross section value. It is easy to check that the larger s/b, the higher the *p*-value and the more precise cross section or resonance parameters determination for a determined statistic level. The value of s/b is particularly relevant for the challenging measurements of radioactive samples with a very small number of atoms available or samples embedded in matrix materials with large contributions from other isotopes [37,38]. In addition, the specific characteristics of the ToF facility, such as instantaneous flux and resolution function, indirectly affect the s/b and must be taken into account for the sensitivity of the experimental setups.

It is important to highlight at this point that evaluated libraries are built from experimental data and, therefore, uncertainties of the original works have a sizable impact on nuclear evaluations and any calculation derived from them either for nuclear astrophysics or applications [15]. Thus, high sensitivity setups are required to improve the state-of-the-art of neutron induced cross sections, and this is in particular the case for the  $(n, \gamma)$  reaction channel. In the next sections we will report on our effort made in the direction of getting better s/b compared to traditionally used C<sub>6</sub>D<sub>6</sub> detection setup.

# 3. Hybrid experimental setup for $(n, \gamma)$ cross section measurements at CERN n\_TOF EAR2

The hybrid setup for  $(n, \gamma)$  cross section measurements, first used in the 2022 campaign, is made of an array of nine sTED units, two conventional large volume  $C_6D_6$ , and one  $LaCl_3(Ce)$  detectors at different angles and distances as shown in Fig. 1. The segmented total-energy detector sTED [22,23] was primarily designed for coping with the extreme count rates characteristic of the high neutron flux at n TOF EAR2. The active volume of each detection unit, aluminum housed, is approximately one-ninth of a liter of  $C_6D_6$  liquid scintillator [19]. In this way, a compact block of  $3 \times 3$  detector modules reaches the overall efficiency of a large-volume C<sub>6</sub>D<sub>6</sub> detector (see Fig. 3 in Ref. [22]). Compared to them, sTED modules also incorporate Photo-Multiplier Tubes (PMTs) Hamamatsu-R11265U, which are designed to withstand high counting rates and provide a fast time response. The use of this PMT helps to mitigate the impact of the  $\gamma$ -flash. The latter is a prompt and intense flux of  $\gamma$ -rays and relativistic particles produced by the arrival of the proton bunch at the spallation target [9]. This intense flux of particles can easily saturate the response of most radiation detectors in the vicinity of the beam-line (see Section 5). Finally, this type of



**Fig. 1.** Pictures of the hybrid  $(n, \gamma)$  experimental setup at EAR2 during the 2022 n\_TOF experimental campaign. The sTED cylindrical configuration surrounds the sample with cylindrical geometry. Two large-volume  $C_6D_6$  detectors are placed in the upstream position, pointing towards the sample at 135°. The inorganic LaCl<sub>3</sub>(Ce) scintillator with a 2-cm thick <sup>6</sup>Li polyethylene shield is also placed at 135° upstream.

PMT also helps to reduce dead-time and pile-up effects (see Ref. [21] for details).

Once the limiting effect of the high count rate was improved thanks to the detector segmentation, the next step was to enhance the detection sensitivity by optimizing the s/b ratio as introduced in Section 2. After a systematic exploration of the background conditions in the neighborhood region of the sample, it became clear that the maximum sensitivity is achieved for as close as possible and equidistant positions of the individual sTED modules with respect to the sample. Thus, instead of using sTED as a compact-block array as proposed in Ref. [23], the individual small-volume detectors were separated and set-up in a cylindrical geometry around the capture sample [39-41] as shown in both panels of Fig. 1. In this configuration detectors avoid any direct interaction with the neutron beam which is important for suppressing undesired neutron interactions with the active volume. The sample under study is placed at the geometrical center of the setup, which coincides with the beam axis, and the distance of detectors from this axis is 4.5 cm, as short as possible to accommodate 9 sTED units.

The inorganic LaCl<sub>2</sub>(Ce) detector with a size of  $50 \times 50 \times 25 \text{ mm}^3$ was placed at an angle of 135° with respect to the beam direction and a distance of 8.5 cm from the sample position. To mitigate neutron sensitivity effects in the LaCl<sub>3</sub>(Ce) detector a 2 cm thick <sup>6</sup>Li-enriched high density polyethylene block, with a size of  $5 \times 5$  cm<sup>2</sup>, was placed in the front of the face of the crystal. The two 1L volume  $C_6D_6$  detectors were also placed at an angle of 135° and a distance of 17 cm from the sample position. The larger distances for all these detectors were needed to compensate their large intrinsic and geometric efficiency in such a way that total count rates per detector were still tolerable and comparable to those of the individual sTED modules. These large-volume detectors enable one to control different important experimental aspects, such as angular distribution effects or details of the capture cascade. In turn, the proper characterization of such effects are vitally important for an accurate assessment of several yield correction factors related to the measuring technique [21].

All detectors were calibrated using standard  $\gamma$ -ray calibration sources, including <sup>137</sup>Cs, <sup>207</sup>Bi, <sup>60</sup>Co, and Am/Be, covering  $\gamma$  rays with energy up to ~4.4 MeV. A common threshold of 200 keV was applied in the data analysis for all detectors. The conversion between ToF and neutron energy was calculated using the relativistic formula [15] with a fixed distance of 19.8 m.

In order to illustrate the sensitivity performance of the sTED compared to the large-volume detectors, a short test using  ${}^{93}Nb(n,\gamma)$  was performed before the  ${}^{94}Nb(n,\gamma)$  cross section measurement campaign [38] using the experimental setup shown in Fig. 1. A full paper with the analysis of both cross sections will be presented in a dedicated publication. Two different sample configurations were measured. The first one corresponded to <sup>93</sup>Nb sample in the form of a disc with 2 mm thickness and 16.34 mm diameter mounted on a 70 µm thick Kapton backing. The second configuration, devoted to background estimation, consisted only of 70 µm thick Kapton backing. These configurations are labeled as *test* and *background*, respectively. The number of protons devoted to each configuration, that in turn are proportional to the neutron beam intensity [9], was  $N' = 1.82 \cdot 10^{16}$  and  $N = 8.77 \cdot 10^{16}$ , respectively. This measurement was used to evaluate the *s*/*b* performance or *p*-value from Eq. (3) for each individual detection system.

The measured count rate spectra normalized to the nominal protonbeam intensity, as a function of neutron energy, are shown in Fig. 2 for the sTED array (a), the LaCl<sub>3</sub>(Ce) detector (b), and the two largevolume  $C_6D_6$  detectors in panel (c) and (d). The *test* and *background configurations* are displayed with red and blue curves, respectively. The neutron-energy range shown in Fig. 2 covers the first two neutron resonances in <sup>93</sup>Nb at 37 and 43 eV, which are well known from previous works [42–44]. The (beam-off) background related to the environmental activity is negligible in all cases and will not be discussed further.

The results obtained for the *background configuration* in all the detection devices (see Fig. 2) indicate that the background level is similar in the neighborhood of the sample position, regardless of the exact detector position or distance, at least within the region of about 20 cm around the beam axis. On the other hand, as expected, the *test configuration* shows a much larger sensitivity to the neutron resonances for the sTED modules than for other detection systems. Indeed, for the allocated N' in the *test configuration*, the two prominent resonances in <sup>93</sup>Nb+n are clearly visible in all detection systems. However, only the sTED array (panel a) allows one to visually identify the small resonance at 40 eV due to a Ta-contaminant in the sample.

The larger sensitivity of the sTED detectors is to be ascribed to the higher s/b ratio, which in turn arises from the inverse-squared sample-to-detector distance dependence of the signal intensity and the approximately constant background level in the sample surroundings. This fact reflect that it is much better a geometric optimization than the use of large active volumes. The relative s/b ratio with respect to sTED in the neutron energy range from 32 eV to 46 eV is approximately 0.24 for the  $C_6D_6$  detectors and 0.35 for the LaCl<sub>3</sub>(Ce) detector. This magnitude was calculated using the neutron resonance integrals and background shown in Fig. 2. The statistical significance of the results for the different detection systems was calculated using the pvalue. The latter indicates the probability of rejecting the hypothesis of background-only, as given by Eqs. (2) and (3). In other words, a *p*-value close to unity indicates a strong confidence that the measured data corresponds to a resonance, and not to a casual background fluctuation. The calculation was performed as a function of the neutron energy





**Fig. 2.** Neutron energy spectra registered by all detection systems during the experiment in the neutron energy range corresponding with the two first neutron resonances of  $^{93}$ Nb. Panels (a), (b), (c) and (d) corresponds to sTED, LaCl<sub>3</sub>(Ce) and both C<sub>6</sub>D<sub>6</sub>  $\gamma$ -ray detection systems, respectively.



**Fig. 3.** The significance of a signal observation, expressed as *p*-value, as a function of the neutron energy for the different experimental devices.

using individual neutron energy bin. The results are shown in Fig. 3 (note the cut-off value at 0.8). Using a *p*-value threshold of 0.95, both the large-volume  $C_6D_6$  and  $LaCl_3(Ce)$  detectors detect a signal different from the background around the resonance energies: 36.5-37.5 eV and 43.0-44.0 eV. However, the *p*-values above-threshold for sTED appear in a significantly wider energy range around the resonance at 40 eV (a Ta contaminant), which is not detected with the other systems.

In short, the results depicted in Figs. 2 and 3 indicate that the measuring time needed to achieve a statistical significance comparable

to the sTED array with both  $C_6D_6$  and LaCl<sub>3</sub>(Ce) would be about 3 times larger. Thus, the sTED array in this geometric configuration can measure  $(n, \gamma)$  cross-sections approximately three times smaller than those measurable by other detection systems, enabling the study of more challenging cross sections at CERN n\_TOF EAR2, such as *s*-process bottlenecks, branching points, or samples with a reduced number of atoms [41]. Apart from detection sensitivity or signal-to-background ratio, another aspect that becomes of importance for a capture experiment is the capture-cascade detection efficiency. This can be best studied from a Monte Carlo perspective, as discussed in the following section.

# 4. Monte Carlo simulation of the hybrid $(n, \gamma)$ setup at n\_TOF EAR2

A Monte Carlo simulation of the hybrid experimental setup was created using the GEANT4 toolkit (version 4.10.7) [29] with the aim of comparing the different detection systems in terms of efficiency. The implemented geometry is illustrated in Fig. 4, which can be compared directly with the pictures of the actual experimental setup in both panels of Fig. 1. The MC simulation includes all detection devices along with their housing materials, quartz optical windows and PMTs at their corresponding experimental positions. Additionally, aluminum and carbon fiber detector holding structures, neutron beam pipes, and the three pillars anchoring the experimental setup present in EAR2 were incorporated into the geometry to make it as close as possible to the reality. For the MC simulations, a modular physics approach was employed to automatically handle different categories of physics, such as electromagnetic (EM), hadronic, and decay processes. In the current code, the modular physics setup includes the standard electromagnetic package option 4, radioactive decay, the Particle High Precision model, and neutron thermal scattering. This MC application, therefore, enables



Fig. 4. Detailed geometry of the experimental setup introduced in Section 3 implemented in GEANT4 simulations. The positions and distances of the  $\gamma$ -ray detection device models correspond to the experiment.



Fig. 5. Calculated MC detection efficiency as a function of incident  $\gamma$ -ray energy for sTED (red), STAR (Orange), individual  $C_6D_6$  (blue) and LaCl<sub>3</sub>(Ce) detectors (green).

accurate simulations of both  $\gamma$ -ray and neutron interactions across a broad range of energies.

The  $\gamma$ -ray detection efficiency curve for each detector was determined by simulating mono-energetic  $\gamma$ -rays isotropically emitted from the sample position assuming a point like source. A common threshold of 200 keV in deposited energy was applied to all detectors similarly to experimental values (see Section 3). The efficiency as a function of the  $\gamma$ -ray energy for the sTED array, the individual large-volume C<sub>6</sub>D<sub>6</sub> detectors, and the LaCl<sub>3</sub>(Ce) detector are displayed in Fig. 5.

Because of the different sample-detector distances at which each detection system can operate in the high-flux conditions of EAR2 (see Section 3), there are notable differences in absolute detection efficiency. The  $\gamma$ -ray efficiency of the sTED array is a factor between 4 and 7 higher than that of a C<sub>6</sub>D<sub>6</sub> detector. A similar ratio is found with respect to the LaCl<sub>3</sub>(Ce) detector. This fact, together with the experimental results discussed in Section 3, reflects the twofold advantage of using small-detection volumes in high neutron flux environments. On the one hand, detector-to-sample distances can be significantly shortened (Fig. 1) thus enhancing s/b ratio (Fig. 3), as well as overall detection efficiency (Fig. 5). On the other hand, a further degree of flexibility in detector geometry is achieved (see Fig. 1), which allows for the additional steps in terms of sensitivity optimization with commonly

used scintillation materials. It is noteworthy that further refinements and optimizations may be also attained by means of new scintillation materials, as it discussed in the following section.

# 5. Future prospects based on solid deuterated-stilbene scintillators

The possibility to utilize solid plastic scintillators for measuring radiative neutron-capture cross sections via the ToF technique has been a topic of discussion for many years. However, the presence of hydrogen in significant amounts in commercially available scintillation materials, like BC-537, has hindered this possibility owing to the high neutron sensitivity and related backgrounds. In the last decade, developments of new solution-growth methodologies for high-quality conventional trans-stilbene organic crystals allowed to prepare high purity deuterated stilbene crystals (stilbene-d12) [27,28]. Their chemical composition ( $C_{10}D_{12}$ ), high density (1.24 g/cm<sup>3</sup>), large  $n/\gamma$  pulse-shape discrimination capabilities, large light yield output, and smaller or negligible chemical risks, at variance with purified deuterated benzenes, makes this scintillators, which have been extensively used for neutron-capture ToF experiments over the last decades.

There are other potential benefits of using solid scintillators. Specifically, active detection volumes in solid (and non hygroscopic) form allow one to use a thinner encapsulation, avoid expansion volumes required with liquid-based detectors, and permits also to eliminate the boro-silicate quartz window, commonly used for optical coupling between the PMT and the liquid scintillator. Finally, a solid scintillator can be more easily coupled to a SiPM photosensor, thereby reducing further the amount of materials in the detection device [45]. All these aspects should further contribute to a reduction of the intrinsic neutron sensitivity of the detection device, thus enabling improved accuracy for the measurement in neutron energy regions with a dominant scattering cross section [6,18].

Empowered by this motivation, we present here a conceptual design study for a future Stilbene-d12 deTector ARray, referred to as STAR, primarily aimed at  $(n, \gamma)$  measurements of very small and/or radioactive samples at CERN n\_TOF EAR2. First experimental results with one STAR-module prototype will then presented and discussed in the following section.

For the design and optimization of STAR we conducted a MC study using a modified GEANT4 implementation of the hybrid setup introduced in Section 4. The modification consisted of replacing the



**Fig. 6.** Graphical representation of the STAR experimental setup as implemented in GEANT4. sTED detectors in Fig. 4 have been replaced by STAR detectors, while the rest of elements, including the structural materials are placed in the same position as for the setup introduced in Section 3.

individual sTED detectors with nine stilbene-d12 detectors. Each individual stilbene crystal has a size of  $25 \times 25 \times 50$  mm<sup>3</sup>, resulting in a total active volume of 281.25 cm<sup>3</sup>. The distance of the face of the individual crystals from the center of the setup was kept the same as for sTED. In this way, one avoids any possible direct interaction with the neutron beam and their performance can be directly compared to other detectors in the hybrid setup. Considering STAR dimensions, this represents a reduction factor of 3.56 in volume compared to sTED while preserving a comparable  $\gamma$ -ray efficiency. Further, in the simulations a 1.5 mm thick SiPM was included instead of a PMT used for sTED, matching the small  $25 \times 25$  mm<sup>2</sup> detector face. The active volume of the individual modules was housed using 1 mm Teflon wrap combined with 0.5 mm aluminum layer. The geometry implementation in the GEANT4 simulation is displayed in Fig. 6.

The  $\gamma$ -ray detection efficiency for STAR was obtained in a similar fashion as described in Section 4 and thus, a direct comparison with the rest of the detection systems is straight forward and shown in Fig. 5. Compared to the large-volume C<sub>6</sub>D<sub>6</sub> and LaCl<sub>3</sub>(Ce) detectors the efficiency of STAR is larger by a factor  $\approx$  4-6×, depending on  $\gamma$ -ray energy. Detection efficiencies of STAR and sTED are thus similar in magnitude but display some difference in the  $E_{\gamma}$  dependence. Although the efficiency of STAR is lower than for sTED for a significant range of  $\gamma$ -ray energy dependence, the total efficiency for actual cascades could be significantly higher, especially for heavy nuclei, where the  $\gamma$ -ray deexcitation is dominantly made via several transitions rather by a direct transition to the ground state.

In order to illustrate this feature, a simulation of the response of individual detector systems was undertaken for s-wave resonances formed in <sup>197</sup>Au(n,  $\gamma$ ) reaction. This reaction is commonly used for yield normalization purposes in neutron-capture ToF experiments [15, 46,47]. The  $\gamma$ -ray cascades for <sup>197</sup>Au(n,  $\gamma$ ) reaction were simulated with *DICEBOX* [48] code. These cascades give a very good reproduction of spectra from C<sub>6</sub>D<sub>6</sub> detectors measured at n\_TOF [21] as well as from DANCE detector [49,50]. The capture efficiency obtained from this simulation for each detection system is displayed in Fig. 7. In summary, the capture-cascade detection efficiency for <sup>197</sup>Au(n,  $\gamma$ ) is expected to be about 10% larger for STAR compared to sTED because of the  $\gamma$ ray cascades pattern. Similar variations, in one direction or another, are expected depending on the exact  $\gamma$ -ray energies emitted during the decay of individual isotope.

# 6. First experimental results with a stilbene-d12 prototype

A first STAR module prototype was assembled, consisting of a small  $25 \times 25 \times 13 \text{ mm}^3$  stilbene-d12 crystal produced at Lawrence



**Fig. 7.** Calculated MC detection efficiency for  ${}^{197}Au(n, \gamma)$  decay for sTED (red), STAR (Orange), individual large-volume C<sub>6</sub>D<sub>6</sub> (blue) and LaCl<sub>3</sub>(Ce) detectors (green).

Livermore National Laboratory [27,28]. The crystal was wrapped with 0.5 mm Teflon and 0.5 mm adhesive black tape. The module was optically coupled to a Hamamatsu-R11265U PMT and tested at the Instituto de Física Corpuscular (IFIC) and later also at CERN n\_TOF EAR2 [51]. The goal of the present prototype and measurements was twofold. On the one hand, these measurements were intended to study and characterize the neutron-gamma Pulse Shape Discrimination (PSD) capabilities of the stilbene-d12 crystal coupled to the aforementioned PMT. On the other hand, the measurements enabled one to assess the performance of this small prototype in a representative ToF experiment, thereby comparing the stilbene-d12 response with respect to large-volume  $C_6D_6$  detectors.

# $6.1. \ {\it Machine-learning} \ {\it aided} \ {\it neutron-gamma} \ {\it pulse-shape} \ {\it discrimination} \ {\it with} \ {\it stilbene-d12}$

The PSD capability of stilbene-d12 was investigated at IFIC using a Lecroy Oscilloscope (64MXi-A), equipped with a bandwidth of 600 MHz and a maximum sampling rate of 10 Gs/s. Two radioactive sources of  $^{22}$ Na and  $^{252}$ Cf were utilized for this aim. The first source was used for energy calibration of the detector and to cross-check the  $\gamma$ -ray pulse shape waveform. The second source, which emits both neutrons and  $\gamma$ -rays, was used to perform pulse shape discrimination for the detector.

The raw detector pulses were digitized at 5 Gs/s and recorded for offline analysis. Before the analysis all registered waveforms were first aligned in time using a constant-fraction digital algorithm. The signal baseline level, calculated as the average of 100 samples before signal, was also subtracted and all the acquired waveforms. All the treated digitized waveforms were normalized in amplitude.

Two different PSD methodologies were investigated, namely, the charge comparison method [52] and the unsupervised Machine Learning *k*-means++ [53–55] algorithm implemented in the scikit-LEARN library [56]. For the charge comparison method, a PSD parameter, *Q*, is defined as the ratio between the fast ( $I_{fast}$ ) and the total ( $I_{total}$ ) waveform integral ( $Q = I_{fast}/I_{total}$ ) [52]. An algorithm for determination of integration time windows was implemented in order to obtain the best value for the Figure of Merit (*FoM*) [52] defined as

$$FoM = \frac{D_n - D_\gamma}{W_n + W_\gamma} \tag{4}$$



**Fig. 8.** WCSS values as a function of number of clusters, *k*. The change in slope determines the optimum number of clusters for the *k*-mean++ algorithm.

where  $D_n$  and  $D_\gamma$  are the mean positions of the neutron and  $\gamma$ -ray Q parameter, respectively, and  $W_n$  and  $W_\gamma$  are the corresponding FWHM. It was found that the best *FoM* was reached for a fast- and total-time integration windows of 20 ns and 100 ns, respectively. Using a minimum deposited energy of 0.2 MeVee one obtains a *FoM* of 1.17. This result is comparable with the results reported previously [27,28]. Therefore, one can conclude that the use of the fast PMT is not affecting noticeably the PSD capability when compared with previous works.

For the unsupervised Machine Learning *k*-means++ algorithm, the set of waveforms  $(\vec{x_1}, \ldots, \vec{x_n})$ , was fed into the algorithm aiming to partition the <sup>252</sup>Cf dataset into *k* clusters. Each waveform is a *d*-dimensional vector and therefore, the clustering algorithm will work in a *d*-dimensional space, corresponding to the length of the digitized waveforms. The algorithm minimizes the within-cluster sum of squares (WCSS) defined as

$$\underset{S}{\operatorname{argmin}} \sum_{i=1}^{k} \sum_{\vec{x} \in S_{i}} \|\vec{x} - \vec{\mu_{i}}\|^{2}$$
(5)

where  $\vec{\mu}_i$  is the *d*-dimensional mean of points within the cluster  $S_i$ . The number of clusters, *k*, is the only parameter to be optimized during the process and it is usually determined with the "elbow" technique [57]. It is a heuristic method consisting of plotting WCSS as a function of *k* as shown in Fig. 8 and identifying the "elbow" point where WCSS changes the slope abruptly. In this particular case, approximately 10<sup>5</sup> waveforms were fed to the algorithm. Each of them had a fixed length of 350 samples covering the range of 70 ns due to memory constrains of the computer used for this work. Fig. 8 clearly indicates that the curve slope changes drastically at *k*=2. In fact, 2 clusters correspond to a physical solution of the problem since there are only two types of particles interacting with the active volume of the detector:  $\gamma$ -rays and neutrons emitted from the <sup>252</sup>Cf calibration source.

In order to compare with the traditional charge comparison PSD Q parameter, the corresponding *k*-means++ derived label ( $\gamma$  or neutron) is plotted together with the Q parameter as a function of the deposited electron equivalent energy in Fig. 9. The labels obtained from the *k*-means++ are presented as red and blue markers, corresponding to  $\gamma$ -and neutron-tagged events, respectively. Identification of  $\gamma$ -rays and neutrons was based on signals from <sup>22</sup>Na.



**Fig. 9.** PSD parameter *Q* derived from charge comparison method as a function of the deposited electron equivalent energy for the small stilbene-d12 module for pulses from the  $^{252}$ Cf. Blue and red dot points indicate neutron and  $\gamma$  signals as derived from *k*-means++ algorithm, respectively.



Fig. 10. Average neutron and  $\gamma$ -ray events waveforms calculated based on the identification obtained from the *k*-means++ approach.

The *k*-means++ results are in good agreement with the charge comparison method. The distribution of points from *k*-means++ looks very reasonable and indicates a possibility to extend particle identification further down in electron equivalent deposited energy. This possibility will be investigated in future works. Such development might provide better discrimination of neutron from  $\gamma$ -ray particles.

For completeness, Fig. 10 shows the average pulse shape experimentally obtained by means of the *k*-means++ approach. The waveforms, with an average pulse time width of only 18 ns and 22 ns FWHM for  $\gamma$ and neutron signals are well suited for high count rate environments, such as the one at n\_TOF EAR2 [21].



Fig. 11. Top view of the experimental setup used during the test performed at n\_TOF EAR2. Two large-volume  $C_6D_6$  detectors, three sTED modules and the stilbene-d12 detector were placed at 90° and different distances (see text) to the center of the neutron beam line.

### 6.2. Characterization measurements at CERN n\_ToF EAR2

The small STAR module prototype was then tested in a dedicated experimental campaign at n\_TOF EAR2 together with several other detectors [51]. An upstream picture of the experimental setup is shown in Fig. 11. Two large-volume  $C_6D_6$  detectors, and three sTED individual modules were placed together with the stilbene-d12 prototype, all of them at 90° with respect to the neutron beam. The distance from each detector to the sample (neutron-beam axis) was adjusted in order to achieve similar count rates in all detectors. All detectors were fully functional even under the largest count-rate conditions reached and the corresponding data could be reliably analyzed and compared. In particular, the large-volume  $C_6D_6$  were placed at 17 cm from the sample, whereas the sTED modules were located at 5 cm distance, a slightly different distance from the nominal one described in Section 3. The stilbene-d12 prototype was placed at the same distance from the beam axis as individual sTED modules.

The time-response function of the stilbene-d12 prototype was compared with the other detectors. The main interest of this measurement was to investigate the time response of the STAR module in realistic conditions. Fig. 12 shows the average pulse shape waveform of all different detectors obtained with an <sup>88</sup>Y calibration source placed at the center of the experimental setup. The risetime, calculated as the time difference between 10% and 90% of the signal maximum is of 3 ns, 4 ns and 8 ns for the stilbene-d12, the sTED and C<sub>6</sub>D<sub>6</sub> detectors, respectively. The pulse time-width (FWHM) for large-volume C<sub>6</sub>D<sub>6</sub>, sTED and stilbene-d12 are 40 ns, 23 ns and 18 ns, respectively. To some extent, the better timing characteristics of the sTED and stilbene-d12 detectors are to be ascribed to their smaller volume and consequently more efficient light collection [21]. Therefore, slightly worse timing characteristics may be found for the future (larger size) stilbene-d12 crystals of 25  $\times$  25  $\times$  50  $mm^3$  planned to be used in STAR (see Section 5). However, the excellent results found here lend confidence that the deuterated stilbene will provide also sufficiently good timeresponse for neutron-capture experiments in the high-flux conditions of EAR2.



Fig. 12. Average pulse shape obtained for all the detectors present during the experimental campaign at n\_TOF EAR2 described in Section 6.2.



Fig. 13. Response of the different detectors to the prompt  $\gamma$ -flash with an empty sample. The inset shows a zoom in the high-neutron energy region.

Another important aspect for any neutron-capture detection setup at EAR2 is the recovery time needed after the prompt  $\gamma$ -flash. Fig. 13 illustrates the recovery from the  $\gamma$ -flash at EAR2 for C<sub>6</sub>D<sub>6</sub>, sTED and stilbene-d12 detectors, covering a ToF region down to a few MeV neutron energy. The figure corresponds to a signal obtained with no sample used, which corresponds to the minimum (ideal)  $\gamma$ -flash response. The large-volume C<sub>6</sub>D<sub>6</sub> detectors recover the baseline level at times corresponding to energies of a few MeV( $\sim$ 7 MeV). On the other hand, the response of the sTED and stilbene-d12 detectors reaches the baseline already at about three times higher neutron energies. This is an important feature of small-volume detectors that allows to measure at higher neutron energies than with the large-volume C<sub>6</sub>D<sub>6</sub> ones.

Finally, for a quick assessment of the signal-to-background ratio with the stilbene-d12 detector a short measurement with a gold sample was performed. Fig. 14 shows the measured count-rate spectra for all types of used detectors. For comparison purposes, the sTED and stilbene-d12 yields were normalized to  $C_6D_6$  spectrum in the saturated region of the 4.9 eV gold resonance. The s/b ratio for both sTED and stilbene-d12 is about a factor 9 better than for the large-volume  $C_6D_6$  detector, when comparing the valley between 4.9 and 60 eV resonances. A detailed inspection of the spectrum measured with stilbene-d12 further revealed no contributions from chemical or isotopic contaminations arising from neutron-capture in the crystal



**Fig. 14.** Experimental yield of  $^{197}$ Au( $n, \gamma$ ) measured at n\_TOF EAR2 with different detection systems normalized to the saturated resonance. Results are shown for a large-volume C<sub>6</sub>D<sub>6</sub> detector, the sTED detectors present and the stilbene-d12 prototype described in Section 6.2.

itself. This result is very important and indicates the quality and purity of the stilbene-d12 crystal used in these tests.

In summary, the results presented above essentially validate the main features (timing performance, background, sensitivity) of the new stilbene-d12 material for neutron-capture time-of-flight experiments, and they pave the path for the future realization of the full STAR array. Further validation measurements are foreseen in the near future utilizing a STAR demonstrator based on stilbene-d12 crystals with a volume of  $25 \times 25 \times 50 \text{ mm}^3$ .

# 7. Conclusions and outlook

Conventional  $C_6D_6$  detectors regularly used for  $(n, \gamma)$  cross-section measurements over the past decades face significant limitations in performance when utilized in high-intensity Time-of-Flight measuring stations, such as EAR2 at the n\_TOF facility. These limitations arise as a consequence of their relatively large volume and readout photosensors. Experiments have shown that such effects can be mitigated by means of an array of small-volume detectors. An array of nine small  $C_6D_6$ modules, known as sTED, represents nowadays the state of the art in this field. Initially designed as a compact block of  $3 \times 3$  sTED modules, in this work we have found that a cylindrical arrangement in a close geometry around the capture sample provides much better performance, particularly for optimizing the *s/b* ratio, which is crucial for neutron-capture cross-section measurements involving small cross sections or samples with very few atoms.

One of the primary objectives of this paper is therefore to present the key features of the sTED setup in cylindrical configuration, derived from measurements using a hybrid configuration for  $(n, \gamma)$  experiments at CERN n\_TOF EAR2 facility in 2022. This setup includes nine sTED detector modules arranged around the capture sample, alongside with two large-volume conventional C<sub>6</sub>D<sub>6</sub> detectors and a LaCl<sub>3</sub>(Ce) detector positioned at 135° relative to the neutron beam. Using data from a short measurement of a <sup>93</sup>Nb sample, we demonstrate a nearly tenfold improvement in the s/b ratio with the sTED in cylindrical configuration compared to conventional C<sub>6</sub>D<sub>6</sub> detectors. However, the contribution of regular C<sub>6</sub>D<sub>6</sub> and a LaCl<sub>3</sub>(Ce) detectors in the hybrid setup is still of relevance towards a better control of the systematic effects related to the angular distribution of  $\gamma$ -rays emitted in low-multiplicity capture cascades [58], as well as other potential systematic effects [13]. The experimental results detailed in Section 3 demonstrate that the sTED array in this cylindrical configuration is sensitive to  $(n, \gamma)$  crosssections approximately three times smaller than those detectable by

other detection devices. Therefore, the use of this experimental setup will enable studies of more challenging cross-sections, such as *s*-process bottlenecks, branching points, or samples with a limited number of available atoms.

Apart from signal-to-background ratio, the efficiency for detecting capture events was explored by means of GEANT4 simulations. The results shown in Section 4 indicate that the efficiency of the sTED array for individual  $\gamma$ -rays is between 4 and 7 times higher than that of a C<sub>6</sub>D<sub>6</sub> detector. In terms of capture-cascade detection efficiency, for the <sup>197</sup>Au( $n, \gamma$ ) reaction (see Fig. 7), the sTED efficiency was found to be 5 to 6 times higher than that of other detection systems.

The GEANT4 model was also utilized to explore alternative detection systems that could provide additional enhancements. Thus, in Section 5 a new detection system called STAR (Stilbene-d12 deTector ARray) was proposed and investigated both conceptually and experimentally. This next generation of neutron-capture detectors is expected to provide enhanced performance and improved chemical safety. The MC simulations indicate a comparable or slightly higher (10%) efficiency compared to sTED for  $(n, \gamma)$  cross section measurements, as shown in Fig. 7. However, one of the main advantages of STAR is expected to come from their smaller (solid) volume, and the larger flexibility in terms of detection-setup optimization and customization for each type of experiment.

To demonstrate the feasibility of the STAR detection setup, we conducted initial measurements using a small stilbene-d12 prototype of  $25 \times 25 \times 13 \text{ mm}^3$  coupled to a fast PMT. The output signals have a width of 18 ns and 22 ns FWHM for  $\gamma$ -ray and neutron events, respectively. As shown in Section 6, the PSD value (1.17) is consistent with previous works. An unsupervised machine learning approach was developed to further explore the PSD at low deposited energies, yielding compatible results. In the future, this approach will be tested at n\_TOF to evaluate the possibility of automatic PSD as a means for further background rejection.

Some of the measurements for the small prototype were performed directly at EAR2, in a configuration that included additional detectors for comparison. As described in Section 6.2, the prototype has a pulse shape faster than the small-volume  $C_6D_6$  detectors, fast recovery from the  $\gamma$ -flash in the MeV region and a signal-to-background ratio of a factor 9 larger compared with large-volume  $C_6D_6$  detectors. In short, the key performance indicators of this small stilbene detector module yield great expectations for future experiments with STAR, involving high-intensity and high-quality neutron beams, both at CERN n\_TOF EAR2 and other high neutron flux facilities.

### CRediT authorship contribution statement

**J. Balibrea-Correa:** Writing – original draft, Software, Methodology, Investigation, Funding acquisition, Formal analysis, Data curation, Conceptualization.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Acknowledgments

The authors acknowledge support from the Spanish Ministerio de Ciencia e Innovación under grants PID2019-104714GB-C21, PID2022-138297NB-C21 and the financial support from MCIN, PCI2022-135037-2 funded by MCIN/AEI/10.13039/501100011033/ and the European Union NextGenerationEU and Generalitat Valenciana in the call PRTR PC I+D+i ASFAE/2022/027. This work was supported by European Union NextGeneration EU/PRTR project C17.I02.P02-SGI\_GICS Nuevas actuaciones en grandes infraestructuras de investigación europeas e internacionales, subproject C17.I02.P02.S01.S03 CSIC CERN. The corresponding author JB acknowledges support from grant ICJ220-045122-I funded by MCIN/AEI/10.13039/501100011033. Author VB is a beneficiary of the Margarita Salas grant (MS21- 178) for the requalification of the Spanish university system from the Ministry of Universities of the Government of Spain, financed by the European Union, NextGenerationUE. Author JL acknowledges the support provided by postdoctoral grants FJC2020-044688-I by MCIN/AEI/10.13039/501100011033 and CIAPOS/2022/020 funded by the Generalitat Valenciana and the European Social Fund and a PhD grant PRE2023 from CSIC. The work on stilbene-d12 development was performed under the auspices of the U.S. Department of Energy by Lawrence Livermore National Laboratory under Contract DE-AC52-07NA27344. Support from the funding agencies of all other participating institutes are also gratefully acknowledged.

### Data availability

Data will be made available on request.

#### References

- R.A. Alpher, H. Bethe, G. Gamow, The origin of chemical elements, Phys. Rev. 73 (1948) 803–804, http://dx.doi.org/10.1103/PhysRev.73.803, URL https: //link.aps.org/doi/10.1103/PhysRev.73.803.
- [2] R.A. Alpher, R.C. Herman, On the relative abundance of the elements, Phys. Rev. 74 (1948) 1737–1742, http://dx.doi.org/10.1103/PhysRev.74.1737, URL https://link.aps.org/doi/10.1103/PhysRev.74.1737.
- [3] R.A. Alpher, A neutron-capture theory of the formation and relative abundance of the elements, Phys. Rev. 74 (1948) 1577–1589, http://dx.doi.org/10.1103/ PhysRev.74.1577, URL https://link.aps.org/doi/10.1103/PhysRev.74.1577.
- [4] R.A. Alpher, R. Herman, G.A. Gamow, Thermonuclear reactions in the expanding universe, Phys. Rev. 74 (1948) 1198–1199, http://dx.doi.org/10.1103/PhysRev. 74.1198.2, URL https://link.aps.org/doi/10.1103/PhysRev.74.1198.2.
- [5] E.M. Burbidge, G.R. Burbidge, W.A. Fowler, F. Hoyle, Synthesis of the elements in stars, Rev. Modern Phys. 29 (1957) 547–650, http://dx.doi.org/10. 1103/RevModPhys.29.547, URL https://link.aps.org/doi/10.1103/RevModPhys. 29.547.
- [6] F. Käppeler, R. Gallino, S. Bisterzo, W. Aoki, The *s* process: Nuclear physics, stellar models, and observations, Rev. Modern Phys. 83 (2011) 157–193, http:// dx.doi.org/10.1103/RevModPhys.83.157, URL https://link.aps.org/doi/10.1103/ RevModPhys.83.157.
- [7] A. Arcones, F.-K. Thielemann, Origin of the elements, Astron. Astrophys. Rev. 31 (1) (2022) 1, http://dx.doi.org/10.1007/s00159-022-00146-x.
- [8] C. Massimi, S. Cristallo, C. Domingo-Pardo, C. Lederer-Woods, N\_TOF: Measurements of key reactions of interest to AGB stars, Universe 8 (2) (2022) http://dx.doi.org/10.3390/universe8020100, URL https://www.mdpi.com/2218-1997/8/2/100.
- [9] E. Chiaveri, O. Aberle, V. Alcayne, S. Amaducci, J. Andrzejewski, L. Audouin, V. Babiano-Suarez, M. Bacak, M. Barbagallo, S. Bennett, E. Berthoumieux, D. Bosnar, A. Brown, M. Busso, M. Caamaño, L. Caballero, M. Calviani, F. Calviño, D. Cano-Ott, A. Casanovas, F. Cerutti, N. Colonna, G. Cortés, M. Cortés-Giraldo, L. Cosentino, S. Cristallo, L. Damone, P. Davies, M. Diakaki, M. Dietz, C. Domingo-Pardo, R. Dressler, Q. Ducasse, E. Dupont, I. Durán, Z. Eleme, B. Fernández-Domíngez, A. Ferrari, I. Ferro-Gonçalves, P. Finocchiaro, V. Furman, R. Garg, A. Gawlik, S. Gilardoni, K. Göbel, E. González-Romero, C. Guerrero, F. Gunsing, S. Heinitz, J. Heyse, D. Jenkins, E. Jericha, U. Jiri, A. Junghans, Y. Kadi, F. Käppeler, A. Kimura, I. Knapová, M. Kokkoris, Y. Kopatch, M. Krticka, D. Kurtulgil, I. Ladarescu, C. Lederer-Woods, J. Lerendegui-Marco, S.-J. Lonsdale, D. Macina, A. Manna, T. Martínez, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, E. Maugeri, A. Mazzone, E. Mendoza, A. Mengoni, V. Michalopoulou, P. Milazzo, M. Millán-Callado, F. Mingrone, J. Moreno-Soto, A. Musumarra, A. Negret, F. Ogállar, A. Oprea, N. Patronis, A. Pavlik, J. Perkowski, C. Petrone, L. Piersanti, E. Pirovano, I. Porras, J. Praena, J. Quesada, D. Ramos Doval, R. Reifarth, D. Rochman, C. Rubbia, M. Sabaté-Gilarte, A. Saxena, P. Schillebeeckx, D. Schumann, A. Sekhar, A. Smith, N. Sosnin, P. Sprung, A. Stamatopoulos, G. Tagliente, J. Tain, A. Tarifeño-Saldivia, L. Tassan-Got, B. Thomas, P. Torres-Sánchez, A. Tsinganis, S. Urlass, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P. Woods, T. Wright, P. Zugec, Status and perspectives of the neutron time-of-flight facility n\_TOF at CERN, EPJ Web Conf. 239 (2020) 17001, http://dx.doi.org/10.1051/epjconf/202023917001.
- [10] A. Bensussan, J. Salome, GELINA: A modern accelerator for high resolution neutron time of flight experiments, Nucl. Instrum. Methods 155 (1) (1978) 11–23, http://dx.doi.org/10.1016/0029-554X(78)90181-7, URL https://www. sciencedirect.com/science/article/pii/0029554X78901817.

- [11] J.-Y. Tang, Q. An, J.-B. Bai, J. Bao, Y. Bao, P. Cao, H.-L. Chen, Q.-P. Chen, Y.-H. Chen, Z. Chen, Z.-Q. Cui, R.-R. Fan, C.-Q. Feng, K.-Q. Gao, X.-L. Gao, M.-H. Gu, C.-C. Han, Z.-J. Han, G.-Z. He, Y.-C. He, Y. Hong, Y.-W. Hu, H.-X. Huang, X.-R. Huang, H.-Y. Jiang, W. Jiang, Z.-J. Jiang, H.-T. Jing, L. Kang, B. Li, C. Li, J.-W. Li, Q. Li, X. Li, Y. Li, J. Liu, R. Liu, S.-B. Liu, X.-Y. Liu, Z. Long, G.-Y. Luan, C.-J. Ning, M.-C. Niu, B.-B. Qi, J. Ren, Z.-Z. Ren, X.-C. Ruan, Z.-H. Song, K. Sun, Z.-J. Sun, Z.-X. Tan, X.-Y. Tang, B.-B. Tian, L.-J. Wang, P.-C. Wang, Z.-H. Wang, Z.-W. Wen, X.-G. Wu, X. Wu, L.-K. Xie, X.-Y. Yang, Y.-W. Yang, H. Yi, L. Yu, T. Yu, Y.-J. Yu, G.-H. Zhang, L.-H. Zhang, Q.-W. Zhang, X.-P. Zhang, Y.-L. Zhang, Z.-Y. Zhang, L.-P. Zhou, Z.-H. Zhou, K.-J. Zhu, Collaboration, The CSNS Back-n, Back-n white neutron source at CSNS and its applications, Nucl. Sci. Tech. 32 (1) (2021) 11, http://dx.doi.org/10.1007/s41365-021-00846-6.
- [12] R. MacKlin, J. Gibbons, Capture-cross-section studies for 30-220-keV neutrons using a new technique, Phys. Rev. 159 (4) (1967) 1007–1012, http://dx.doi.org/10.1103/PhysRev.159.1007, URL https://www.scopus.com/ inward/record.uri?eid=2-s2.0-0000079592&doi=10.1103%2fPhysRev.159. 1007&partnerID=40&md5=d93882b4dc264832db4a05db63e8cfd4. Cited by: 233
- [13] U. Abbondanno, G. Aerts, F. Alvarez-Velarde, H. Álvarez-Pol, S. Andriamonje, J. Andrzejewski, G. Badurek, P. Baumann, F. Bečvář, J. Benlliure, E. Berthoumieux, F. Calviño, D. Cano-Ott, R. Capote, P. Cennini, V. Chepel, E. Chiaveri, N. Colonna, G. Cortes, D. Cortina, A. Couture, J. Cox, S. Dababneh, M. Dahlfors, S. David, R. Dolfini, C. Domingo-Pardo, I. Duran, M. Embid-Segura, L. Ferrant, A. Ferrari, R. Ferreira-Marques, H. Frais-Koelbl, W. Furman, I. Goncalves, R. Gallino, E. Gonzalez-Romero, A. Goverdovski, F. Gramegna, E. Griesmaver, F. Gunsing, B. Haas, R. Haight, M. Heil, A. Herrera-Martinez, S. Isaev, E. Jericha, F. Käppeler, Y. Kadi, D. Karadimos, M. Kerveno, V. Ketlerov, P. Koehler, V. Konovalov, M. Krtička, C. Lamboudis, H. Leeb, A. Lindote, I. Lopes, M. Lozano, S. Lukic, J. Marganiec, S. Marrone, J. Martinez-Val, P. Mastinu, A. Mengoni, P.M. Milazzo, A. Molina-Coballes, C. Moreau, M. Mosconi, F. Neves, H. Oberhummer, S. O'Brien, J. Pancin, T. Papaevangelou, C. Paradela, A. Pavlik, P. Pavlopoulos, J.M. Perlado, L. Perrot, M. Pignatari, R. Plag, A. Plompen, A. Plukis, A. Poch, A. Policarpo, C. Pretel, J. Quesada, S. Raman, W. Rapp, T. Rauscher, R. Reifarth, M. Rosetti, C. Rubbia, G. Rudolf, P. Rullhusen, J. Salgado, J.C. Soares, C. Stephan, G. Tagliente, J. Tain, L. Tassan-Got, L. Tavora, R. Terlizzi, G. Vannini, P. Vaz, A. Ventura, D. Villamarin, M.C. Vincente, V. Vlachoudis, F. Voss, H. Wendler, M. Wiescher, K. Wisshak, n\_TOF Collaboration Collaboration, Neutron capture cross section measurement of <sup>151</sup>Sm at the CERN neutron time of flight facility (n tof), Phys. Rev. Lett. 93 (2004) 161103, http://dx.doi.org/10.1103/PhysRevLett.93.161103, URL https://link.aps.org/doi/10.1103/PhysRevLett.93.161103.
- [14] A. Borella, G. Aerts, F. Gunsing, M. Moxon, P. Schillebeeckx, R. Wynants, The use of C<sub>6</sub>D<sub>6</sub> detectors for neutron induced capture cross-section measurements in the resonance region, Nucl. Instrum. Methods Phys. Res. A 577 (2007) 626–640, http://dx.doi.org/10.1016/j.nima.2007.03.034.
- [15] P. Schillebeeckx, B. Becker, Y. Danon, K. Guber, H. Harada, J. Heyse, A. Junghans, S. Kopecky, C. Massimi, M. Moxon, N. Otuka, I. Sirakov, K. Volev, Determination of resonance parameters and their covariances from neutron induced reaction cross section data, Nucl. Data Sheets 113 (12) (2012) 3054–3100, http://dx.doi.org/10.1016/j.nds.2012.11.005, URL https://www.sciencedirect.com/science/article/pii/S0090375212000919. Special Issue on Nuclear Reaction Data.
- [16] J. Ren, X. Ruan, J. Bao, G. Luan, W. Jiang, Q. An, H. Bai, P. Cao, Q. Chen, Y. Chen, P. Cheng, Z. Cui, R. Fan, C. Feng, M. Gu, F. Guo, C. Han, Z. Han, G. He, P. Zhu, The c6d<sub>6</sub> detector system on the back-n beam line of CSNS, Radiat. Detect. Technol. Methods 3 (2019) http://dx.doi.org/10.1007/s41605-019-0129-8.
- [17] F. Corvi, A. Prevignano, H. Liskien, P. Smith, An experimental method for determining the total efficiency and the response function of a γ-ray detector in the range 0.5–10 MeV, Nucl. Instrum. Methods Phys. Res. A 265 (3) (1988) 475–484, http://dx.doi.org/10.1016/S0168-9002(98)90016-X, URL https: //www.sciencedirect.com/science/article/pii/S016890029890016X.
- [18] R. Plag, M. Heil, F. Käppeler, P. Pavlopoulos, R. Reifarth, K. Wisshak, An optimized C6d6 detector for studies of resonance-dominated (n,y) cross-sections, Nucl. Instrum. Methods Phys. Res. A 496 (2) (2003) 425–436, http://dx.doi.org/ 10.1016/S0168-9002(02)01749-7, URL https://www.sciencedirect.com/science/ article/pii/S0168900202017497.
- [19] P. Mastinu, R. Baccomi, E. Berthoumieux, D. Cano-Ott, F. Gramegna, C. Guerrero, C. Massimi, P.M. Milazzo, F. Mingrone, J. Praena, G. Prete, A.R. García, New C<sub>6</sub>D<sub>6</sub> detectors: reduced neutron sensitivity and improved safety, 2013, URL https://cds.cern.ch/record/1558147.
- [20] C. Weiß, E. Chiaveri, S. Girod, V. Vlachoudis, O. Aberle, S. Barros, I. Bergström, E. Berthoumieux, M. Calviani, C. Guerrero, M. Sabaté-Gilarte, A. Tsinganis, J. Andrzejewski, L. Audouin, M. Bacak, J. Balibrea-Correa, M. Barbagallo, V. Bécares, C. Beinrucker, F. Belloni, F. Bečvář, J. Billowes, D. Bosnar, M. Brugger, M. Caamaño, F. Calviño, D. Cano-Ott, F. Cerutti, N. Colonna, G. Cortés, M. Cortés-Giraldo, L. Cosentino, L. Damone, K. Deo, M. Diakaki, C. Domingo-Pardo, E. Dupont, I. Durán, R. Dressler, B. Fernández-Domínguez, A. Ferrari, P. Ferreira, P. Finocchiaro, R. Frost, V. Furman, S. Ganesan, A. Gheorghe, T. Glodariu, K. Göbel, I. Gonçalves, E. González-Romero, A. Goverdovski, E. Griesmayer, F. Gunsing, H. Harada, T. Heftrich, S. Heinitz, A. Hernández-Prieto, J. Heyse, D. Jenkins, E. Jericha, Y. Kadi, F. Käppeler, T. Katabuchi, P. Kavrigin, V. Ketlerov, V. Khryachkov, A. Kimura, N. Kivel, M. Kokkoris, M.

Krtička, E. Leal-Cidoncha, C. Lederer, H. Leeb, J. Lerendegui, M. Licata, S.L. Meo, D. López, R. Losito, D. Macina, J. Marganiec, T. Martínez, C. Massimi, P. Mastinu, M. Mastromarco, F. Matteucci, E. Mendoza, A. Mengoni, P. Milazzo, F. Mingrone, M. Mirea, S. Montesano, A. Musumarra, R. Nolte, R. Palomo Pinto, C. Paradela, N. Patronis, A. Pavlik, J. Perkowski, I. Porras, J. Praena, J. Quesada, T. Rauscher, R. Reifarth, A. Riego-Perez, M. Robles, C. Rubbia, J. Ryan, A. Saxena, P. Schillebeeckx, S. Schmidt, D. Schumann, P. Sedyshev, G. Smith, A. Stamatopoulos, P. Steinegger, S. Suryanarayana, G. Tagliente, J. Tain, A. Tarifeño-Saldivia, L. Tassan-Got, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, R. Vlastou, A. Wallner, S. Warren, M. Weigand, T. Wright, P. Žugec, The new vertical neutron beam line at the CERN n\_TOF facility design and outlook on the performance, Nucl. Instrum. Methods Phys. Res. A 799 (2015) 90–98, http://dx.doi.org/10.1016/j.nima.2015.07.027, URL https://www.sciencedirect.com/science/article/pii/S0168900215008566.

- [21] J. Balibrea-Correa, J. Lerendegui-Marco, V. Babiano-Suarez, C. Domingo-Pardo, I. Ladarescu, A. Tarifeño-Saldivia, G. de la Fuente-Rosales, V. Alcayne, D. Cano-Ott, E. González-Romero, T. Martínez, E. Mendoza, A.P. de Rada, J.P. del Olmo, A. Sánchez-Caballero, A. Casanovas, F. Calviño, S. Valenta, O. Aberle, S. Altieri, S. Amaducci, J. Andrzejewski, M. Bacak, C. Beltrami, S. Bennett, A. Bernardes, E. Berthoumieux, R. Beyer, M. Boromiza, D. Bosnar, M. Caamaño, M. Calviani, D. Castelluccio, F. Cerutti, G. Cescutti, S. Chasapoglou, E. Chiaveri, P. Colombetti, N. Colonna, P.C. Camprini, G. Cortés, M. Cortés-Giraldo, L. Cosentino, S. Cristallo, S. Dellmann, M. Di Castro, S. Di Maria, M. Diakaki, M. Dietz, R. Dressler, E. Dupont, I. Durán, Z. Eleme, S. Fargier, B. Fernández, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, V. Furman, F. García-Infantes, A. Gawlik-Ramikega, G. Gervino, S. Gilardoni, C. Guerrero, F. Gunsing, C. Gustavino, J. Heyse, W. Hillman, D. Jenkins, E. Jericha, A. Junghans, Y. Kadi, K. Kaperoni, G. Kaur, A. Kimura, I. Knapová, M. Kokkoris, Y. Kopatch, M. Krtička, N. Kyritsis, C. Lederer-Woods, G. Lerner, A. Manna, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, E. Maugeri, A. Mazzone, A. Mengoni, V. Michalopoulou, P. Milazzo, R. Mucciola, F. Murtas, E. Musacchio-Gonzalez, A. Musumarra, A. Negret, P. Pérez-Maroto, N. Patronis, J. Pavón-Rodríguez, M. Pellegriti, J. Perkowski, C. Petrone, E. Pirovano, S. Pomp, I. Porras, J. Praena, J. Ouesada, R. Reifarth, D. Rochman, Y. Romanets, C. Rubbia, M. Sabaté-Gilarte, P. Schillebeeckx, D. Schumann, A. Sekhar, A. Smith, N. Sosnin, M. Stamati, A. Sturniolo, G. Tagliente, D. Tarrío, P. Torres-Sánchez, E. Vagena, V. Variale, P. Vaz, G. Vecchio, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P. Woods, T. Wright, R. Zarrella, P. Žugec, Pushing the high count rate limits of scintillation detectors for challenging neutroncapture experiments, Nucl. Instrum. Methods Phys. Res. A 1064 (2024) 169385, http://dx.doi.org/10.1016/j.nima.2024.169385, URL https://www.sciencedirect. com/science/article/pii/S0168900224003115.
- [22] V. Alcayne, D. Cano-Ott, J. Garcia, E. González-Romero, T. Martínez, E. Mendoza, A. Sánchez, J. Plaza, J. Balibrea-Correa, A. Casanovas, C. Domingo-Pardo, J. Lerendegui-Marco, O. Aberle, S. Altieri, S. Amaducci, H. Amar Es-Sghir, J. Andrzejewski, V. Babiano-Suarez, M. Bacak, J. Balibrea, S. Bennett, A.P. Bernardes, E. Berthoumieux, D. Bosnar, M. Busso, M. Caamaño, F. Calviño, M. Calviani, D. Cano-Ott, A. Casanovas, D.M. Castelluccio, F. Cerutti, G. Cescutti, S. Chasapoglou, E. Chiaveri, P. Colombetti, N. Colonna, P.C. Console Camprini, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, S. Cristallo, M. Di Castro, D. Diacono, M. Diakaki, M. Dietz, C. Domingo-Pardo, R. Dressler, E. Dupont, I. Durán, Z. Eleme, S. Fargier, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, V. Furman, F. García-Infantes, A. Gawlik-Ramiega, G. Gervino, S. Gilardoni, E. González-Romero, C. Guerrero, F. Gunsing, C. Gustavino, J. Heyse, D.G. Jenkins, E. Jericha, A. Junghans, Y. Kadi, T. Katabuchi, I. Knapová, M. Kokkoris, Y. Kopatch, M. Krticka, D. Kurtulgil, I. Ladarescu, C. Lederer-Woods, J. Lerendegui-Marco, G. Lerner, A. Manna, T. Martínez, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, F. Matteucci, E.A. Maugeri, A. Mazzone, E. Mendoza, A. Mengoni, V. Michalopoulou, P.M. Milazzo, R. Mucciola, F. Murtas+, E. Musacchio-Gonzalez, A. Musumarra, A. Negret, A. Oprea, P. Pérez-Maroto, N. Patronis, J.A. Pavón-Rodríguez, M.G. Pellegriti, J. Perkowski, C. Petrone, L. Piersanti, E. Pirovano, S. Pomp, I. Porras, J. Praena, N. Protti, J.M. Quesada, T. Rauscher, R. Reifarth, D. Rochman, Y. Romanets, F. Romano, C. Rubbia, A. Sánchez, M. Sabaté-Gilarte, P. Schillebeeckx, D. Schumann, A. Sekhar, A.G. Smith, N.V. Sosnin, M. Spelta, M.E. Stamati, G. Tagliente, A. Tarifeño-Saldivia, D. Tarrío, N. Terranova, P. Torres-Sánchez, S. Urlass, S. Valenta, V. Variale, P. Vaz, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P.J. Woods, T. Wright, P. Zugec, the n\_TOF Collaboration, A segmented total energy detector (sTED) for  $(n,\gamma)$  cross section measurements at n\_TOF EAR2, in: EPJ Web Conf., vol. 284, 2023, p. 01043, http://dx.doi.org/10.1051/epjconf/202328401043.
- [23] V. Alcayne, D. Cano-Ott, J. Garcia, E. González-Romero, T. Martínez, A.P. de Rada, J. Plaza, A. Sánchez-Caballero, J. Balibrea-Correa, C. Domingo-Pardo, J. Lerendegui-Marco, A. Casanovas, F. Calviño, O. Aberle, S. Altieri, S. Amaducci, J. Andrzejewski, V. Babiano-Suarez, M. Bacak, C. Beltrami, S. Bennett, A. Bernardes, E. Berthoumieux, R. Beyer, M. Boromiza, D. Bosnar, M. Caamaño, M. Calviani, D. Castelluccio, F. Cerutti, G. Cescutti, S. Chasapoglou, E. Chiaveri, P. Colombetti, N. Colonna, P.C. Camprini, G. Cortés, M. Cortés-Giraldo, L. Cosentino, S. Dellmann, M. Di Castro, S. Di Maria, M. Diakaki, M. Dietz, R. Dressler, E. Dupont, I. Durán, Z. Eleme, S. Fargier, B. Fernández, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, V. Furman, F. García-Infantes, A. Gawlik-Ramiega, G. Gervino, S. Gilardoni, C. Guerrero, F. Gunsing, C. Gustavino, J. Heyse, W. Hillman, D. Jenkins, E.

Jericha, A. Junghans, Y. Kadi, K. Kaperoni, G. Kaur, A. Kimura, J. Knapová, M. Kokkoris, Y. Kopatch, M. Krtička, N. Kyritsis, I. Ladarescu, C. Lederer-Woods, G. Lerner, A. Manna, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, E. Maugeri, A. Mazzone, E. Mendoza, A. Mengoni, V. Michalopoulou, P. Milazzo, R. Mucciola F Murtas F Musacchio-Gonzalez A Musumarra A Negret P Pérez-Maroto, N. Patronis, J. Pavón-Rodríguez, M. Pellegriti, J. Perkowski, C. Petrone, L. Piersanti, E. Pirovano, S. Pomp, I. Porras, J. Praena, J. Quesada, R. Reifarth, D. Rochman, Y. Romanets, C. Rubbia, M. Sabaté-Gilarte, P. Schillebeeckx, D. Schumann, A. Sekhar, A. Smith, N. Sosnin, M. Spelta, M. Stamati, A. Sturniolo, G. Tagliente, A. Tarifeño-Saldivia, D. Tarrío, P. Torres-Sánchez, E. Vagena, S. Valenta, V. Variale, P. Vaz, G. Vecchio, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P. Woods, T. Wright, R. Zarrella, P. Žugec, A segmented total energy detector (sTED) optimized for  $(n, \gamma)$  cross-section measurements at n TOF EAR2, Radiat. Phys. Chem. 217 (2024) 111525, http://dx.doi.org/10.1016/j. radphyschem.2024.111525, URL https://www.sciencedirect.com/science/article/ pii/S0969806X24000173.

- [24] V. Babiano-Suárez, J. Lerendegui-Marco, J. Balibrea-Correa, L. Caballero, D. Calvo, I. Ladarescu, D. Real, C. Domingo-Pardo, F. Calviño, A. Casanovas, A. Tarifeño-Saldivia, V. Alcavne, C. Guerrero, M.A. Millán-Callado, T. Rodríguez-González, M. Barbagallo, O. Aberle, S. Amaducci, J. Andrzejewski, L. Audouin, M. Bacak, S. Bennett, E. Berthoumieux, J. Billowes, D. Bosnar, A. Brown, M. Busso, M. Caamaño, M. Calviani, D. Cano-Ott, F. Cerutti, E. Chiaveri, N. Colonna, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, S. Cristallo, L.A. Damone, P.J. Davies, M. Diakaki, M. Dietz, R. Dressler, Q. Ducasse, E. Dupont, I. Durán, Z. Eleme, B. Fernández-Domínguez, A. Ferrari, P. Finocchiaro, V. Furman, K. Göbel, R. Garg, A. Gawlik, S. Gilardoni, I.F. Gonçalves, E. González-Romero, F. Gunsing, H. Harada, S. Heinitz, J. Heyse, D.G. Jenkins, A. Junghans, F. Käppeler, Y. Kadi, A. Kimura, I. Knapova, M. Kokkoris, Y. Kopatch, M. Krtička, D. Kurtulgil, C. Lederer-Woods, H. Leeb, S.J. Lonsdale, D. Macina, A. Manna, T. Martinez, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, E.A. Maugeri, A. Mazzone, E. Mendoza, A. Mengoni, V. Michalopoulou, P.M. Milazzo, F. Mingrone, J. Moreno-Soto, A. Musumarra, A. Negret, F. Ogállar, A. Oprea, N. Patronis, A. Pavlik, J. Perkowski, L. Persanti, C. Petrone, E. Pirovano, I. Porras, J. Praena, J.M. Quesada, D. Ramos-Doval, T. Rauscher, R. Reifarth, D. Rochman, C. Rubbia, M. Sabaté-Gilarte, A. Saxena, P. Schillebeeckx, D. Schumann, A. Sekhar, A.G. Smith, N.V. Sosnin, P. Sprung, A. Stamatopoulos, G. Tagliente, J.L. Tain, L. Tassan-Got, T. Thomas, P. Torres-Sánchez, A. Tsinganis, J. Ulrich, S. Urlass, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P.J. Woods, T. Wright, P. Žugec, Imaging neutron capture cross sections: i-TED proof-of-concept and future prospects based on machine-learning techniques, Eur. Phys. J. A 57 (6) (2021) 197, http://dx.doi.org/10.1140/epja/s10050-021-00507-7.
- [25] C. Domingo-Pardo, I-TED: A novel concept for high-sensitivity (n,γ) cross-section measurements, Nucl. Instrum. Methods Phys. Res. A 825 (2016) 78–86, http: //dx.doi.org/10.1016/j.nima.2016.04.002, URL https://www.sciencedirect.com/ science/article/pii/S0168900216301619.
- [26] E. Mendoza, V. Alcayne, D. Cano-Ott, E. González-Romero, T. Martínez, A. Pérez de Rada, A. Sánchez-Caballero, J. Balibrea-Correa, C. Domingo-Pardo, J. Lerendegui-Marco, F. Calviño, C. Guerrero, Neutron capture measurements with high efficiency detectors and the pulse height weighting technique, Nucl. Instrum. Methods Phys. Res. A 1047 (2023) 167894, http://dx.doi.org/10. 1016/j.nima.2022.167894, URL https://www.sciencedirect.com/science/article/pii/\$01689002201186X.
- [27] F. Becchetti, R. Raymond, R. Torres-Isea, A. Di Fulvio, S. Clarke, S. Pozzi, M. Febbraro, Recent developments in deuterated scintillators for neutron measurements at low-energy accelerators, Nucl. Instrum. Methods Phys. Res. A 874 (2017) 72–78, http://dx.doi.org/10.1016/j.nima.2017.08.034, URL https: //www.sciencedirect.com/science/article/pii/S0168900217309051.
- [28] M. Carman, A. Glenn, A. Mabe, F. Becchetti, S. Payne, N. Zaitseva, Solution growth of a deuterated trans-stilbene crystal for fast neutron detection, J. Cryst. Growth 498 (2018) 51–55, http://dx.doi.org/10.1016/j.jcrysgro.2018.05.021, URL https://www.sciencedirect.com/science/article/pii/S0022024818302483.
- [29] J. Allison, K. Amako, J. Apostolakis, P. Arce, M. Asai, T. Aso, E. Bagli, A. Bagulya, S. Banerjee, G. Barrand, B. Beck, A. Bogdanov, D. Brandt, J. Brown, H. Burkhardt, P. Canal, D. Cano-Ott, S. Chauvie, K. Cho, G. Cirrone, G. Cooperman, M. Cortés-Giraldo, G. Cosmo, G. Cuttone, G. Depaola, L. Desorgher, X. Dong, A. Dotti, V. Elvira, G. Folger, Z. Francis, A. Galoyan, L. Garnier, M. Gayer, K. Genser, V. Grichine, S. Guatelli, P. Guèye, P. Gumplinger, A. Howard, I. Hřivnáčová, S. Hwang, S. Incerti, A. Ivanchenko, V. Ivanchenko, F. Jones, S. Jun, P. Kaitaniemi, N. Karakatsanis, M. Karamitros, M. Kelsey, A. Kimura, T. Koi, H. Kurashige, A. Lechner, S. Lee, F. Longo, M. Maire, D. Mancusi, A. Mantero, E. Mendoza, B. Morgan, K. Murakami, T. Nikitina, L. Pandola, P. Paprocki, J. Perl, I. Petrović, M. Pia, W. Pokorski, J. Quesada, M. Raine, M. Reis, A. Ribon, A. Ristić Fira, F. Romano, G. Russo, G. Santin, T. Sasaki, D. Sawkey, J. Shin, I. Strakovsky, A. Taborda, S. Tanaka, B. Tomé, T. Toshito, H. Tran, P. Truscott, L. Urban, V. Uzhinsky, J. Verbeke, M. Verderi, B. Wendt, H. Wenzel, D. Wright, D. Wright, T. Yamashita, J. Yarba, H. Yoshida, Recent developments in Geant4, Nucl. Instrum. Methods Phys. Res. A 835 (2016) 186-225, http://dx.doi.org/10.1016/j.nima.2016.06.125, URL http://www.sciencedirect.com/science/article/pii/S0168900216306957.

- [30] N.M. Chiera, E.A. Maugeri, I. Danilov, J. Balibrea-Correa, C. Domingo-Pardo, U. Köster, J. Lerendegui-Marco, M. Veicht, I. Zivadinovic, D. Schumann, Preparation of PbSe targets for <sup>79</sup>se neutron capture cross section studies, Nucl. Instrum. Methods Phys. Res. A 1029 (2022) 166443, http://dx.doi.org/10. 1016/j.nima.2022.166443, URL https://www.sciencedirect.com/science/article/ pii/S0168900222000882.
- [31] J. Lerendegui-Marco, C. Guerrero, E. Mendoza, J.M. Quesada, K. Eberhardt, A.R. Junghans, M. Krtička, O. Aberle, J. Andrzeiewski, L. Audouin, V. Bécares, M. Bacak, J. Balibrea, M. Barbagallo, S. Barros, F. Bečvář, C. Beinrucker, E. Berthoumieux, J. Billowes, D. Bosnar, M. Brugger, M. Caamaño, F. Calviño, M. Calviani, D. Cano-Ott, R. Cardella, A. Casanovas, D.M. Castelluccio, F. Cerutti, Y.H. Chen, E. Chiaveri, N. Colonna, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, L.A. Damone, M. Diakaki, M. Dietz, C. Domingo-Pardo, R. Dressler, E. Dupont, I. Durán, B. Fernández-Domínguez, A. Ferrari, P. Ferreira, P. Finocchiaro, V. Furman, K. Göbel, A.R. García, A. Gawlik, T. Glodariu, I.F. Gonçalves, E. González-Romero, A. Goverdovski, E. Griesmayer, F. Gunsing, H. Harada, T. Heftrich, S. Heinitz, J. Heyse, D.G. Jenkins, E. Jericha, F. Käppeler, Y. Kadi, T. Katabuchi, P. Kavrigin, V. Ketlerov, V. Khryachkov, A. Kimura, N. Kivel, M. Kokkoris, E. Leal-Cidoncha, C. Lederer, H. Leeb, S. Lo Meo, S.J. Lonsdale, R. Losito, D. Macina, J. Marganiec, T. Martínez, C. Massimi, P. Mastinu, M. Mastromarco, F. Matteucci, F.A. Maugeri, A. Mengoni, P.M. Milazzo, F. Mingrone, M. Mirea, S. Montesano, A. Musumarra, R. Nolte, A. Oprea, N. Patronis, A. Pavlik, J. Perkowski, J.I. Porras, J. Praena, K. Rajeev, T. Rauscher, R. Reifarth, A. Riego-Perez, P.C. Rout, C. Rubbia, J.A. Ryan, M. Sabaté-Gilarte, A. Saxena, P. Schillebeeckx, S. Schmidt, D. Schumann, P. Sedyshev, A.G. Smith, A. Stamatopoulos, G. Tagliente, J.L. Tain, A. Tarifeño-Saldivia, L. Tassan-Got, A. Tsinganis, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, V. Vlachoudis, R. Vlastou, A. Wallner, S. Warren, M. Weigand, C. Weiss, C. Wolf, P.J. Woods, T. Wright, P. Žğec, n\_TOF Collaboration Collaboration, Radiative neutron capture on 242Pu in the resonance region at the CERN n\_tof-EAR1 facility, Phys. Rev. C 97 (2018) 024605, http://dx.doi.org/10.1103/PhysRevC.97.024605, URL https: //link.aps.org/doi/10.1103/PhysRevC.97.024605.
- [32] J. Balibrea-Correa, E. Mendoza, D. Cano-Ott, E. González, R. Capote, M. Krtička, S. Altstadt, J. Andrzejewski, L. Audouin, V. Bécares, M. Barbagallo, F. Bečvář, F. Belloni, E. Berthoumieux, J. Billowes, V. Boccone, D. Bosnar, M. Brugger, M. Calviani, F. Calviño, C. Carrapiço, F. Cerutti, E. Chiaveri, M. Chin, N. Colonna, G. Cortés, M.A. Cortés-Giraldo, M. Diakaki, C. Domingo-Pardo, R. Dressler, I. Durán, C. Eleftheriadis, A. Ferrari, K. Fraval, V. Furman, K. Göbel, M.B. Gómez-Hornillos, S. Ganesan, A.R. García, G. Giubrone, I.F. Gonçalves, A. Goverdovski, E. Griesmayer, C. Guerrero, F. Gunsing, T. Heftrich, A. Hernández-Prieto, J. Hevse, D.G. Jenkins, E. Jericha, F. Käppeler, Y. Kadi, D. Karadimos, T. Katabuchi, V. Ketlerov, V. Khryachkov, N. Kivel, P. Koehler, M. Kokkoris, J. Kroll, C. Lampoudis, C. Langer, E. Leal-Cidoncha, C. Lederer, H. Leeb, L.S. Leong, J. Lerendegui-Marco, R. Losito, A. Mallick, A. Manousos, J. Marganiec, T. Martínez, C. Massimi, P. Mastinu, M. Mastromarco, A. Mengoni, P.M. Milazzo, F. Mingrone, M. Mirea, W. Mondelaers, C. Paradela, A. Pavlik, J. Perkowski, A.J.M. Plompen, J. Praena, J.M. Quesada, T. Rauscher, R. Reifarth, A. Riego-Perez, M. Robles, C. Rubbia, J.A. Ryan, M. Sabaté-Gilarte, R. Sarmento, A. Saxena, P. Schillebeeckx, S. Schmidt, D. Schumann, P. Sedyshev, G. Tagliente, J.L. Tain, A. Tarifeño-Saldivia, D. Tarrío, L. Tassan-Got, A. Tsinganis, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, M.J. Vermeulen, V. Vlachoudis, R. Vlastou, A. Wallner, T. Ware, M. Weigand, C. Weiss, T. Wright, P. Žugec, n\_TOF Collaboration Collaboration, Measurement of the  $\alpha$  ratio and  $(n, \gamma)$  cross section of <sup>235</sup>U from 0.2 to 200 ev at n TOF, Phys. Rev. C 102 (2020) 044615, http://dx.doi.org/10.1103/PhysRevC. 102.044615, URL https://link.aps.org/doi/10.1103/PhysRevC.102.044615.
- [33] C. Guerrero, J. Lerendegui-Marco, M. Paul, M. Tessler, S. Heinitz, C. Domingo-Pardo, S. Cristallo, R. Dressler, S. Halfon, N. Kivel, U. Köster, E.A. Maugeri, T. Palchan-Hazan, J.M. Quesada, D. Rochman, D. Schumann, L. Weissman, O. Aberle, S. Amaducci, J. Andrzejewski, L. Audouin, V. Bécares, M. Bacak, J. Balibrea, A. Barak, M. Barbagallo, S. Barros, F. Bečvář, C. Beinrucker, D. Berkovits, E. Berthoumieux, J. Billowes, D. Bosnar, M. Brugger, Y. Buzaglo, M. Caamaño, F. Calviño, M. Calviani, D. Cano-Ott, R. Cardella, A. Casanovas, D.M. Castelluccio, F. Cerutti, Y.H. Chen, E. Chiaveri, N. Colonna, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, H. Dafna, A. Damone, M. Diakaki, M. Dietz, E. Dupont, I. Durán, Y. Eisen, B. Fernández-Domínguez, A. Ferrari, P. Ferreira, P. Finocchiaro, V. Furman, K. Göbel, A.R. García, A. Gawlik, T. Glodariu, I.F. Gonçalves, E. González-Romero, A. Goverdovski, E. Griesmayer, F. Gunsing, H. Harada, T. Heftrich, J. Heyse, T. Hirsh, D.G. Jenkins, E. Jericha, F. Käppeler, Y. Kadi, B. Kaizer, T. Katabuchi, P. Kavrigin, V. Ketlerov, V. Khryachkov, D. Kijel, A. Kimura, M. Kokkoris, A. Kriesel, M. Krtička, E. Leal-Cidoncha, C. Lederer-Woods, H. Leeb, S. Lo Meo, S.J. Lonsdale, R. Losito, D. Macina, A. Manna, J. Marganiec, T. Martínez, C. Massimi, P. Mastinu, M. Mastromarco, F. Matteucci, E. Mendoza, A. Mengoni, P.M. Milazzo, M.A. Millán-Callado, F. Mingrone, M. Mirea, S. Montesano, A. Musumarra, R. Nolte, A. Oprea, N. Patronis, A. Pavlik, J. Perkowski, L. Piersanti, I. Porras, J. Praena, K. Rajeev, T. Rauscher, R. Reifarth, T. Rodríguez-González, P.C. Rout, C. Rubbia, J.A. Ryan, M. Sabaté-Gilarte, A. Saxena, P. Schillebeeckx, S. Schmidt, A. Shor, P. Sedyshev, A.G. Smith, A. Stamatopoulos, G. Tagliente, J.L. Tain, A. Tarifeño-Saldivia, L. Tassan-Got, A. Tsinganis, S. Valenta, G. Vannini, V. Variale, P. Vaz, A. Ventura, V. Vlachoudis, R. Vlastou, A. Wallner, S. Warren, M. Weigand, C. Weiss, C. Wolf, P.J. Woods,

T. Wright, P. Žugec, n\_TOF Collaboration Collaboration, Neutron capture on the *s*-process branching point <sup>171</sup>Tm via time-of-flight and activation, Phys. Rev. Lett. 125 (2020) 142701, http://dx.doi.org/10.1103/PhysRevLett.125.142701, URL https://link.aps.org/doi/10.1103/PhysRevLett.125.142701.

- [34] B. Avanzi, G. Taylor, B. Wong, A. Xian, Modelling and understanding count processes through a Markov-modulated non-homogeneous Poisson process framework, European J. Oper. Res. 290 (1) (2021) 177–195, http: //dx.doi.org/10.1016/j.ejor.2020.07.022, URL https://www.sciencedirect.com/ science/article/pii/S0377221720306330.
- [35] T.P. Li, Y.Q. Ma, Analysis methods for results in gamma-ray astronomy, Astrophys. J. 272 (1983) 317–324, http://dx.doi.org/10.1086/161295.
- [36] G. Vianello, The significance of an excess in a counting experiment: Assessing the impact of systematic uncertainties and the case with a Gaussian background, Astrophys. J. Suppl. Ser. 236 (1) (2018) 17, http://dx.doi.org/10.3847/1538-4365/aab780.
- [37] V. Babiano, J. Balibrea-Correa, L. Caballero, F. Calviño, D. Cano-Ott, A. Casanovas, N. Colonna, S. Cristallo, C. Domingo-Pardo, R. Dressler, E. González, C. Guerrero, S. Heinitz, I. Ladarescu, C. Lederer-Woods, J. Lerendegui-Marco, E.A. Maugeri, E. Mendoza, A. Mengoni, I. Moench, T. Rauscher, N. Sosnin, D. Schumann, A. Tarifeño-Saldivia, n\_TOF Collaboration, The, First measurement of the s-process branching <sup>79</sup>Se(n,γ), Tech. Rep., CERN, Geneva, 2020, URL https://cds.cern.ch/record/2731962.
- [38] V. Babiano, J. Balibrea-Correa, L. Caballero, C. Domingo-Pardo, J. Lerendegui-Marco, First measurement of the <sup>94</sup>Nb(n,γ) cross-section, Tech. Rep., CERN, Geneva, 2020, URL https://cds.cern.ch/record/2731959.
- [39] J. Balibrea-Correa, V. Babiano-Suárez, J. Lerendegui-Marco, C. Domingo-Pardo, I. Ladarescu, A. Tarifeño-Saldivia, V. Alcayne, D. Cano-Ott, E. González-Romero, T. Martínez, E. Mendoza, C. Guerrero, F. Calviño, A. Casanovas, U. Köster, N.M. Chiera, R. Dressler, E.A. Maugeri, D. Schumann, O. Aberle, S. Altieri, S. Amaducci, J. Andrzejewski, M. Bacak, C. Beltrami, S. Bennett, A.P. Bernardes, E. Berthoumieux, R. Beyer, M. Boromiza, D. Bosnar, M. Caamaño, M. Calviani, F. Cerutti, G. Cescutti, F. Chiaveri, P. Colombetti, N. Colonna, P. Console Camprini, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, S. Cristallo, S. Dellmann, M. Di Castro, S. Di Maria, M. Diakaki, M. Dietz, E. Dupont, I. Durán, Z. Eleme, S. Fargier, B. Fernández, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, F. García-Infantes, A. Gawlik-Ramiega, G. Gervino, S. Gilardoni, F. Gunsing, C. Gustavino, J. Heyse, W. Hillman, D.G. Jenkins, E. Jericha, A. Junghans, Y. Kadi, K. Kaperoni, G. Kaur, A. Kimura, I. Knapová, M. Kokkoris, M. Krtička, N. Kyritsis, C. Lederer-Woods, G. Lerner, A. Manna, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, A. Mazzone, A. Mengoni, V. Michalopoulou, P.M. Milazzo, R. Mucciola, F. Murtas, E. Musacchio-Gonzalez, A. Musumarra, A. Negret, A. Pérez de Rada, P. Pérez-Maroto, N. Patronis, J.A. Pavón-Rodríguez, M.G. Pellegriti, J. Perkowski, C. Petrone, E. Pirovano, J. Plaza, S. Pomp, I. Porras, J. Praena, J.M. Quesada, R. Reifarth, D. Rochman, Y. Romanets, C. Rubbia, A. Sánchez-Caballero, M. Sabaté-Gilarte, P. Schillebeeckx, A. Sekhar, A.G. Smith, N.V. Sosnin, M.E. Stamati, A. Sturniolo, G. Tagliente, D. Tarrío, P. Torres-Sánchez, E. Vagena, S. Valenta, V. Variale, P. Vaz, G. Vecchio, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P.J. Woods, T. Wright, R. Zarrella, P. Žugec, First measurement of the  ${}^{94}Nb(n,\gamma)$  cross section at the CERN n TOF facility, EPJ Web Conf. 279 (2023) 06004, http://dx.doi.org/10.1051/epjconf/ 202327906004.
- [40] J. Lerendegui-Marco, V. Babiano-Suá rez, J. Balibrea-Correa, C. Domingo-Pardo, I. Ladarescu, A. Tarifeño-Saldivia, V. Alcayne, D. Cano-Ott, E. González-Romero, T. Martínez, E. Mendoza, C. Guerrero, F. Calviño, A. Casanovas, U. Köster, N.M. Chiera, R. Dressler, E.A. Maugeri, D. Schumann, O. Aberle, S. Altieri, S. Amaducci, J. Andrzejewski, M. Bacak, C. Beltrami, S. Bennett, A.P. Bernardes, E. Berthoumieux, R. Bever, M. Boromiza, D. Bosnar, M. Caamaño, M. Calviani, F. Cerutti, G. Cescutti, E. Chiaveri, P. Colombetti, N. Colonna, P.C. Camprini, G. Cortés, M.A. Cortés-Giraldo, L. Cosentino, S. Cristallo, S. Dellmann, M. Di Castro, S. Di Maria, M. Diakaki, M. Dietz, E. Dupont, I. Durán, Z. Eleme, S. Fargier, B. Fernández, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, F. García-Infantes, A. Gawlik-Ramięga, G. Gervino, S. Gilardoni, F. Gunsing, C. Gustavino, J. Heyse, W. Hillman, D.G. Jenkins, E. Jericha, A. Junghans, Y. Kadi, K. Kaperoni, G. Kaur, A. Kimura, I. Knapová, M. Kokkoris, M. Krtička, N. Kyritsis, C. Lederer-Woods, G. Lerner, A. Manna, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, A. Mazzone, A. Mengoni, V. Michalopoulou, P.M. Milazzo, R. Mucciola, F. Murtas, E. Musacchio-Gonzalez, A. Musumarra, A. Negret, A.P. de Rada, P. Pérez-Maroto, N. Patronis, J.A. Pavón-Rodríguez, M.G. Pellegriti, J. Perkowski, C. Petrone, E. Pirovano, J. Plaza, S. Pomp, I. Porras, J. Praena, J.M. Quesada, R. Reifarth, D. Rochman, Y. Romanets, C. Rubbia, A. Sánchez-Caballero, M. Sabaté-Gilarte, P. Schillebeeckx, A. Sekhar, A.G. Smith, N.V. Sosnin, M.E. Stamati, A. Sturniolo, G. Tagliente, D. Tarrío, P. Torres-Sánchez, E. Vagena, S. Valenta, V. Variale, P. Vaz, G. Vecchio, D. Vescovi, V. Vlachoudis, R. Vlastou, A. Wallner, P.J. Woods, T. Wright, R. Zarrella, P. Žugec, New detection systems for an enhanced sensitivity in key stellar (n,y) measurements, in: S. Freeman, C. Lederer-Woods, A. Manna, A. Mengoni (Eds.), in: EPJ Web of Conferences, vol. 279, EDP Sciences, 2023, p. 13001, http://dx.doi.org/10.1051/epjconf/202327913001.
- [41] C. Domingo-Pardo, V. Babiano-Suarez, J. Balibrea-Correa, L. Caballero, I. Ladarescu, J. Lerendegui-Marco, J.L. Tain, A. Tarifeño-Saldivia, O. Aberle, V. Alcayne, S. Altieri, S. Amaducci, J. Andrzejewski, M. Bacak, C. Beltrami, S.

Bennett, A.P. Bernardes, F. Berthoumieux, M. Boromiza, D. Bosnar, M. Caamaño, F. Calviño, M. Calviani, D. Cano-Ott, A. Casanovas, F. Cerutti, G. Cescutti, S. Chasapoglou, E. Chiaveri, N.M. Chiera, P. Colombetti, N. Colonna, P.C. Camprini, G. Cortés, M. A.Cortés-Giraldo, L. Cosentino, S. Cristallo, S. Dellmann, M. Di Castro S Di Maria M Diakaki M Dietz B Dressler E Dupont I Durán Z. Eleme, S. Fargier, B. Fernández, B. Fernández-Domínguez, P. Finocchiaro, S. Fiore, F. García-Infantes, A. Gawlik-Ramięga, G. Gervino, S. Gilardoni, E. González-Romero, C. Guerrero, F. Gunsing, C. Gustavino, J. Heyse, W. Hillman, D.G. Jenkins, E. Jericha, A. Junghans, Y. Kadi, K. Kaperoni, F. Käppeler, G. Kaur, A. Kimura, I. Knapová, U. Köster, M. Kokkoris, M. Krtička, N. Kyritsis, C. Lederer-Woods, G. Lerner, A. Manna, T. Martínez, A. Masi, C. Massimi, P. Mastinu, M. Mastromarco, E.A. Maugeri, A. Mazzone, E. Mendoza, A. Mengoni, P.M. Milazzo, I. Mönch, R. Mucciola, F. Murtas, F. Musacchio-Gonzalez, A. Musumarra, A. Negret, A. Pérez de Rada, P. Pérez-Maroto, N. Patronis, J.A. Pavón-Rodríguez, M.G. Pellegriti, J. Perkowski, C. Petrone, E. Pirovano, J. Plaza, S. Pomp, I. Porras, J. Praena, J.M. Quesada, R. Reifarth, D. Rochman, Y. Romanets, C. Rubbia, A. Sánchez, M. Sabaté-Gilarte, P. Schillebeeckx, D. Schumann, A. Sekhar, A.G. Smith, N.V. Sosnin, M. Stamati, A. Sturniolo, G. Tagliente, D. Tarrío, P. Torres-Sánchez, J. Turko, S. Urlass, E. Vagena, S. Valenta, V. Variale, P. Vaz, G. Vecchio, D. Vescovi, V. Vlachoudis, R. Vlastou, T. Wallner, P.J. Woods, T. Wright, R. Zarrella, P. Žugec, Collaboration, The n\_TOF, Advances and new ideas for neutron-capture astrophysics experiments at CERN n TOF. Eur. Phys. J. A 59 (1) (2023) 8, http://dx.doi.org/10.1140/epja/s10050-022-00876-7.

- [42] N.J. Drindak, J.A. Burke, G. Leinweber, J.A. Helm, J.G. Hoole, R.C. Block, Y. Danon, R.E. Slovacek, B.E. Moretti, C.J. Werner, M.E. Overberg, S.A. Kolda, T.M. J., D.P. Barry, Neutron capture and transmission measurements and resonance parameter analysis of niobium, Nucl. Sci. Eng. 154 (3) (2006) 294–301, http://dx.doi.org/10.13182/NSE06-A2634.
- [43] T. Wang, G. Kim, M.-W. Lee, K.-S. Kim, M.-H. Cho, H.-S. Kang, W. Namkung, Measurements of neutron total cross-sections and resonance parameters of niobium using pulsed neutrons generated by an electron linac, J. Radioanal. Nucl. Chem. 289 (3) (2011) 945–952, http://dx.doi.org/10.1007/s10967-011-1216-6.
- [44] S. Endo, A. Kimura, S. Nakamura, O. Iwamoto, N. Iwamoto, G. Rovira, K. Terada, S.-i. Meigo, Y.M. Toh, M. Segawa, Maeda, M. Tsuneyama, Neutron capture and total cross-section measurements and resonance parameter analysis of niobium-93 below 400 eV, J. Nucl. Sci. Technol. 59 (3) (2022) 318–333, http://dx.doi.org/10.1080/00223131.2021.1970040.
- [45] J. Balibrea-Correa, J. Lerendegui-Marco, D. Calvo, L. Caballero, V. Babiano, I. Ladarescu, M.L. Redondo, J. Tain, A. Tolosa, C. Domingo-Pardo, F. Calvino, A. Casanovas, A. Tarifeño-Saldivia, V. Alcayne, D. Cano-Ott, T. Martinez, C. Guerrero, M. Barbagallo, D. Macina, M. Bacak, A first prototype of C<sub>6</sub>D<sub>6</sub> total-energy detector with SiPM readout for neutron capture time-of-flight experiments, Nucl. Instrum. Methods Phys. Res. A 985 (2021) 164709, http: //dx.doi.org/10.1016/j.nima.2020.164709, URL https://www.sciencedirect.com/science/article/pii/S0168900220311062.
- [46] R.L. Macklin, J. Halperin, R.R. Winters, Absolute neutron capture yield calibration, Nucl. Instrum. Methods 164 (1) (1979) 213–214, http://dx.doi.org/ 10.1016/0029-554X(79)90457-9, URL https://www.sciencedirect.com/science/ article/pii/0029554X79904579.
- [47] C. Lederer, N. Colonna, C. Domingo-Pardo, F. Gunsing, F. Käppeler, C. Massimi, A. Mengoni, A. Wallner, U. Abbondanno, G. Aerts, H. Álvarez, F. Álvarez-Velarde, S. Andriamonje, J. Andrzejewski, P. Assimakopoulos, L. Audouin, G. Badurek, M. Barbagallo, P. Baumann, F. Bečvář, F. Belloni, E. Berthoumieux, M. Calviani, F. Calviño, D. Cano-Ott, R. Capote, C. Carrapiço, A. Carrillo de Albornoz, P. Cennini, V. Chepel, F. Chiaveri, G. Cortes, A. Couture, J. Cox, M. Dahlfors, S. David, I. Dillmann, R. Dolfini, W. Dridi, I. Duran, C. Eleftheriadis, M. Embid-Segura, L. Ferrant, A. Ferrari, R. Ferreira-Marques, L. Fitzpatrick, H. Frais-Koelbl, K. Fujii, W. Furman, I. Goncalves, E. González-Romero, A. Goverdovski, F. Gramegna, E. Griesmayer, C. Guerrero, B. Haas, R. Haight, M. Heil, A. Herrera-Martinez, M. Igashira, S. Isaev, E. Jericha, Y. Kadi, D. Karadimos, D. Karamanis, M. Kerveno, V. Ketlerov, P. Koehler, V. Konovalov, E. Kossionides, M. Krtička, C. Lampoudis, H. Leeb, A. Lindote, I. Lopes, R. Losito, M. Lozano, S. Lukic, J. Marganiec, L. Marques, S. Marrone, T. Martínez, P. Mastinu, E. Mendoza, P.M. Milazzo, C. Moreau, M. Mosconi, F. Neves, H. Oberhummer, S. O'Brien, M. Oshima, J. Pancin, C. Papachristodoulou, C. Papadopoulos, C. Paradela, N. Patronis, A. Pavlik, P. Pavlopoulos, L. Perrot, M.T. Pigni, R. Plag, A. Plompen, A. Plukis, A. Poch, J. Praena, C. Pretel, J. Ouesada, T. Rauscher, R. Reifarth, M. Rosetti, C. Rubbia, G. Rudolf, P. Rullhusen, J. Salgado, C. Santos, L. Sarchiapone, R. Sarmento, I. Savvidis, C. Stephan, G. Tagliente, J.L. Tain, D. Tarrío, L. Tassan-Got, L. Tavora, R. Terlizzi, G. Vannini, P. Vaz, A. Ventura, D. Villamarin, V. Vlachoudis, R. Vlastou, F. Voss, S. Walter, H. Wendler, M. Wiescher, K. Wisshak, n TOF Collaboration Collaboration,  $^{197}Au(n, \gamma)$  cross section in the unresolved resonance region, Phys. Rev. C 83 (2011) 034608, http://dx.doi.org/10.1103/PhysRevC.83.034608, URL https://link.aps.org/doi/10.1103/PhysRevC.83.034608.

- [48] F. Bečvář, Simulation of γ cascades in complex nuclei with emphasis on assessment of uncertainties of cascade-related quantities, Nucl. Instrum. Methods Phys. Res. A 417 (2) (1998) 434–449, http://dx.doi.org/10.1016/ S0168-9002(98)00787-6, URL https://www.sciencedirect.com/science/article/ pii/S0168900298007876.
- [49] M. Heil, R. Reifarth, M. Fowler, R. Haight, F. Käppeler, R. Rundberg, E. Seabury, J. Ullmann, J. Wilhelmy, K. Wisshak, A 4πBaF2 detector for (n,γ) cross-section measurements at a spallation neutron source, Nucl. Instrum. Methods Phys. Res. A 459 (1) (2001) 229–246, http://dx.doi.org/10.1016/S0168-9002(00)00993-1, URL https://www.sciencedirect.com/science/article/pii/S0168900200009931.
- [50] R. Reifarth, T. Bredeweg, A. Alpizar-Vicente, J. Browne, E.-I. Esch, U. Greife, R. Haight, R. Hatarik, A. Kronenberg, J. O'Donnell, R. Rundberg, J. Ullmann, D. Vieira, J. Wilhelmy, J. Wouters, Background identification and suppression for the measurement of (n,γ) reactions with the DANCE array at LANSCE, Nucl. Instrum. Methods Phys. Res. A 531 (3) (2004) 530–543, http://dx.doi. org/10.1016/j.nima.2004.05.096, URL https://www.sciencedirect.com/science/ article/pii/S0168900204011131.
- [51] O. Aberle, V. Alcayne, M. Bacak, J. Balibrea-Correa, N. Colonna, D. Cano-Ott, A. Casanovas, C. Domingo-Pardo, O. Fjeld, F. Gunsing, J. Lerendegui-Marco, C. Lederer-Woods, C. Massimi, E. Mendoza, A. Mengoni, A. Manna, A. Musumarra, N. Patronis, M.G. Pellegriti, Development of new solid-state total-energy detectors for neutron-capture measurements at CERN n\_TOF, Tech. Rep, CERN, Geneva, 2023, URL https://cds.cern.ch/record/2856350.
- [52] M. Sénoville, F. Delaunay, M. Pârlog, N. Achouri, N. Orr, Neutron-γ discrimination with organic scintillators: Intrinsic pulse shape and light yield contributions, Nucl. Instrum. Methods Phys. Res. A 971 (2020) 164080, http://dx.doi.org/10. 1016/j.nima.2020.164080, URL https://www.sciencedirect.com/science/article/ pii/S0168900220305039.
- [53] S.P. Lloyd, Least squares quantization in PCM. Technical Report RR-5497, Tech. Rep, Bell labs, 1957.
- [54] J.B. MacQueen, Some methods for classification and analysis of MultiVariate observations, in: L.M.L. Cam, J. Neyman (Eds.), in: Proc. of the Fifth Berkeley Symposium on Mathematical Statistics and Probability, vol. 1, University of California Press, 1967, pp. 281–297.
- [55] C.M. Bishop, Pattern Recognition and Machine Learning (Information Science and Statistics), Springer-Verlag, Berlin, Heidelberg, 2006.
- [56] F. Pedregosa, G. Varoquaux, A. Gramfort, V. Michel, B. Thirion, O. Grisel, M. Blondel, P. Prettenhofer, R. Weiss, V. Dubourg, J. Vanderplas, A. Passos, D. Cournapeau, M. Brucher, M. Perrot, E. Duchesnay, Scikit-learn: Machine learning in Python, J. Mach. Learn. Res. 12 (2011) 2825–2830.
- [57] R.L. Thorndike, Who belongs in the family? Psychometrika 18 (4) (1953) 267–276, http://dx.doi.org/10.1007/BF02289263.
- C. Domingo-Pardo, U. Abbondanno, G. Aerts, H. Álvarez-Pol, F. Alvarez-Velarde, [58] S. Andriamonje, J. Andrzejewski, P. Assimakopoulos, L. Audouin, G. Badurek, P. Baumann, F. Becvár, E. Berthoumieux, S. Bisterzo, F. Calviño, D. Cano-Ott, R. Capote, C. Carrapiço, P. Cennini, V. Chepel, E. Chiaveri, N. Colonna, G. Cortes, A. Couture, J. Cox, M. Dahlfors, S. David, I. Dillman, R. Dolfini, W. Dridi, I. Duran, C. Eleftheriadis, M. Embid-Segura, L. Ferrant, A. Ferrari, R. Ferreira-Marques, L. Fitzpatrick, H. Frais-Koelbl, K. Fujii, W. Furman, R. Gallino, I. Goncalves, E. Gonzalez-Romero, A. Goverdovski, F. Gramegna, E. Griesmayer, C. Guerrero, F. Gunsing, B. Haas, R. Haight, M. Heil, A. Herrera-Martinez, M. Igashira, S. Isaev, E. Jericha, Y. Kadi, F. Käppeler, D. Karamanis, D. Karadimos, M. Kerveno, V. Ketlerov, P. Koehler, V. Konovalov, E. Kossionides, M. Krtička, C. Lamboudis, H. Leeb, A. Lindote, I. Lopes, M. Lozano, S. Lukic, J. Marganiec, S. Marrone, P. Mastinu, A. Mengoni, P.M. Milazzo, C. Moreau, M. Mosconi, F. Neves, H. Oberhummer, M. Oshima, S. O'Brien, J. Pancin, C. Papachristodoulou, C. Papadopoulos, C. Paradela, N. Patronis, A. Pavlik, P. Pavlopoulos, L. Perrot, R. Plag, A. Plompen, A. Plukis, A. Poch, C. Pretel, J. Quesada, T. Rauscher, R. Reifarth, M. Rosetti, C. Rubbia, G. Rudolf, P. Rullhusen, J. Salgado, L. Sarchiapone, I. Savvidis, C. Stephan, G. Tagliente, J.L. Tain, L. Tassan-Got, L. Tavora, R. Terlizzi, G. Vannini, P. Vaz, A. Ventura, D. Villamarin, M.C. Vincente, V. Vlachoudis, R. Vlastou, F. Voss, S. Walter, H. Wendler, M. Wiescher, K. Wisshak, Resonance capture cross section of Pb207, Phys. Rev. C 74 (5) (2006) 055802, http://dx.doi.org/10.1103/PhysRevC.74.055802, arXiv: nucl-ex/0610039.